FISCAL YEAR 2006-07 ADMINISTRATION REPORT FOR COMMUNITY FACILITIES DISTRICT NO. 6 OF POWAY UNIFIED SCHOOL DISTRICT

June 26, 2006

FISCAL YEAR 2006-07 ADMINISTRATION REPORT FOR COMMUNITY FACILITIES DISTRICT No. 6

PREPARED FOR

POWAY UNIFIED SCHOOL DISTRICT

13626 Twin Peaks Road Poway, California 92064 (858) 748-0010

PREPARED BY

DAVID TAUSSIG & ASSOCIATES, INC.

1301 Dove Street, Suite 600 Newport Beach, California 92660 (949) 955-1500

TABLE OF CONTENTS

<u>Sec</u>	etion	Page
Inti	RODUCTION	1
I.	ANNUAL SPECIAL TAX LEVY FOR FISCAL YEAR 2005-06	2
II.	FUNDS AND ACCOUNTS	3
	A. Sources and Uses of Funds	
	B. Fund and Account Balances	4
III.	SPECIAL TAX REQUIREMENT FOR FISCAL YEAR 2006-07	6
IV.	SPECIAL TAX CLASSIFICATIONS AND DEVELOPMENT UPDATE	7
	A. Special Tax Classifications	7
	B. Development Update	7
V.	METHOD OF APPORTIONMENT	8

EXHIBITS

Exhibit A: Summary of Transactions within Funds and Accounts Fiscal Year 2005-06 **Exhibit B:** Annual Special Tax Rates for Fiscal Year 2006-07

INTRODUCTION

This Administration Report ("Report") provides an analysis of the financial and administrative obligations of Community Facilities District ("CFD") No. 6 of the Poway Unified School District ("District") resulting from the sale of (i) the Series 2002 Special Tax Bonds ("2002 Bonds") issued in September of 2002 in the amount of \$25,000,000 and (ii) the Series 2005 Special Tax Bonds ("2005 Bonds") issued in November of 2005 in the amount of \$44,305,000. All capitalized terms herein are used as defined in the Rate and Method of Apportionment ("RMA"), the Original Bond Indenture ("Indenture") by and between the District and Zions First National Bank, successor to State Street Bank and Trust Company of California, N.A., ("Fiscal Agent") dated August 1, 2002, and the First Supplemental Bond Indenture ("Supplemental Indenture") by and between the District and the Fiscal Agent dated November 1, 2005 for CFD No. 6.

The 2002 Bonds and 2005 Bonds of CFD No. 6 are secured and repaid through the annual levy and collection of Special Taxes from all property subject to the Annual Special Tax within CFD No. 6. In calculating the Annual Special Tax levy for Fiscal Year 2006-07, the Report describes (i) the remaining financial obligations of CFD No. 6 for Fiscal Year 2005-06, (ii) the financial obligations of CFD No. 6 for Fiscal Year 2006-07, and (iii) the amount of new development which has occurred within the boundaries of the CFD.

The Report is organized into the following sections:

Section I

Section I provides a description of the Annual Special Tax levy for Fiscal Year 2005-06, including any delinquent Annual Special Taxes.

Section II

Section II examines the financial activity in the funds and accounts established pursuant to the Indenture and the First Supplemental Indenture. A Fiscal Year summary is provided which lists all disbursements, Annual Special Tax receipts, and interest earnings.

Section III

Section III calculates the Special Tax Requirement based on the financial obligations of CFD No. 6 for Fiscal Year 2006-07.

Section IV

Section IV provides (i) a description of the Special Tax classifications and (ii) an update of the development which has occurred within CFD No. 6.

Section V

Section V describes the Annual Special Tax rates and reviews the methodology used to apportion the Special Tax Requirement between Developed Property and Undeveloped Property and lists the Annual Special Tax rates for Fiscal Year 2006-07.

I. ANNUAL SPECIAL TAX LEVY FOR FISCAL YEAR 2005-06

The aggregate Annual Special Tax levy for Fiscal Year 2005-06 equaled \$3,603,560.90, of which the entire amount was levied on Developed Property. A summary of the levy is shown in Table 1.

TABLE 1

AVERAGE SPECIAL TAX RATES AND LEVY
OF FISCAL YEAR 2005-06

Unit Type	Number of Units	Average Annual Tax	Total Annual Special Taxes
Detached Unit	1,744	\$2,066.26	\$3,603,560.90
Attached Unit	0	\$0.00	\$0.00
Total	1,744	NA	\$3,603,560.90

David Taussig & Associates, Inc. ("DTA") has received delinquency information from the County of San Diego ("County") for the first installment of Fiscal Year 2005-06. The Special Tax delinquencies for the first installment equaled \$54,406.73, resulting in a delinquency rate of 1.51 percent. DTA has concluded that CFD No. 6 will be able to meet its remaining obligations in Fiscal Year 2005-06.

II. FUNDS AND ACCOUNTS

The Indenture established nine (9) funds for CFD No. 6. They are the Special Tax Fund, Bond Fund, Rebate Fund, Redemption Fund, School Facilities Fund, Reserve Fund, Administration Expense Fund, Prepayment Fund, and Cost of Issuance Fund. In addition, within the Bond Fund, an Interest Account and Principal Account were created and within the School Facilities Fund, a 4S Ranch Middle School Hold Back Account was created.

All receipts from the collection of Annual Special Taxes are deposited in the Special Tax Fund, which is administered by the Fiscal Agent. Pursuant to the Indenture, the Fiscal Agent transfers the Special Tax receipts to the funds in the amount and priority set forth below.

- **1. Administrative Expense Fund** an amount equal to the Administrative Expense Requirement for that Fiscal Year;
- **2. Bond Fund** an amount, taking into account any amount then on deposit, sufficient to meet the principal, premiums, if any, and interest due on the next interest payment date;
- **Reserve Fund** an amount required to bring the balance in the Reserve Fund to the Reserve Requirement
- **4. Rebate Fund** an amount required to be set aside or rebated to the Internal Revenue Service; and
- **5. Administrative Expense Fund** any amount, in excess of the Administrative Expense Requirement, necessary to defray the costs of administering CFD No. 6 for such Fiscal Year.

Any monies remaining in the Special Tax Fund after each of the above transfers are made shall remain on deposit in the Special Tax Fund. However, after such time as the levy of Special Taxes on Developed property exceeds the Special Tax Requirement, then excess funds shall be paid to the District to be used to construct school facilities.

The Reserve Requirement is an amount equal to the lesser of (i) ten percent (10%) of the original proceeds of the Bonds, (ii) Maximum Annual Debt Service on the Bonds, or (iii) one hundred twenty-five percent (125%) of the average annual debt service on the Bonds. As of the date of the Report, the Reserve Requirement was \$2,342,556.70 for the 2002 Bonds and \$3,994,281.53 for the 2005 Bonds.

A. Sources and Uses of Funds

Sources of funds for CFD No. 6 during the period of July 1, 2005, to May 1, 2006, totaled \$51,056,426.86. Of this amount, Special Tax receipts totaled \$2,341,663.22, interest earnings equaled \$694,870.52, and \$48,019,893.15 was for the initial deposit of bond proceeds. Tables 2 and 3 show initial deposit of proceeds from the 2005 Bonds and the interest earnings for each fund and account of CFD No. 6 for the period stated above, respectively.

TABLE 2
BOND PROCEEDS
NOVEMBER 22, 2005

Funds	Earnings
Costs of Issuance Fund	\$247,000.00
Reserve Fund	\$3,901,513.25
School Facilities Fund	\$39,057,240.90
Total	\$43,205,754.15

TABLE 3 INTEREST EARNINGS JULY 1, 2005 – MAY 1, 2006

Funds and Accounts	Earnings
Administrative Expense Fund	\$1,436.01
Interest Account	\$486.53
Principal Account	\$0.00
Cost of Issuance Fund	\$973.42
School Facilities Fund	\$424,281.94
Rebate Fund	\$0.00
Redemption Fund	\$0.00
Reserve Fund	\$133,830.57
4S Ranch Middle School Hold Back Account	\$5,541.54
Bond Service Fund	\$0.00
Special Tax Fund	\$77,779.60
Prepayment Account	\$50,540.91
Total	\$694,870.52

Total uses of funds for this period totaled \$24,488,792.96. Of this amount, \$1,952,640.12 in interest was paid to bondholders, \$55,000.00 was paid to the principal of the 2002 Bonds, \$20,682,945.62 was used for the construction of school facilities, \$67,111.81 was paid for legal, consulting, and professional service fees, \$214,030.77 was used for cost of issuance, and \$1,517,064.64 was used for construction.

A detailed analysis of all transactions within these funds and accounts for the period is included as Exhibit A.

B. Account Balances

The balances as of May 1, 2006, in the funds and accounts established pursuant to the Indenture are shown in Table 4.

TABLE 4
FUND AND ACCOUNT BALANCES
AS OF MAY 1, 2006

Funds and Accounts	Balance
Administrative Expense Fund	\$42,216.49
Bond Service Fund	\$0.00
Interest Account	\$98,98
Principal Account	\$0.00
Cost of Issuance Account	\$33,942.65
School Facilities Fund	\$18,798,773.69
Rebate Fund	\$0.00
Redemption Fund	\$0.00
Reserve Fund	\$6,315,491.62
Special Tax Fund	\$3,278,650.71
Prepayment Account	4,864,456.41
4S Ranch Middle School Hold Back Account	\$1.25
Total	\$33,333,631.80

III. SPECIAL TAX REQUIREMENT FOR FISCAL YEAR 2006-07

The Special Tax Requirement of CFD No. 6 for Fiscal Year 2006-07 is \$5,256,325.64. This amount represents (i) interest and principal payments due to bondholders in Calendar Year 2007, (ii) anticipated Administrative Expenses for Fiscal Year 2006-07, and (iii) funding for school facilities. Table 5 shows the calculation of the Special Tax Requirement of CFD No. 6 for Fiscal Year 2006-07.

TABLE 5

ANNUAL SPECIAL TAX REQUIREMENT
FOR FISCAL YEAR 2006-07

FOR FISCAL YEAR 2000-07									
FY 2005-06 Current and Projected Funds		\$4,641,627.64							
Balance of Special Tax Fund (as of 5/01/06)	\$3,278,650.71								
Balance of Interests Account (as of 5/01/06)	\$98.98								
Projected Special Tax Receipts	\$1,297,877.95								
Reimbursement of Special Taxes (from IA A of CFD No. 6)	\$65,000.00								
FY 2005-06 Remaining Obligations		\$4,641,627.64							
Series 2002 Bonds									
Interest Due September 1, 2006	\$672,470.00								
Principal Due September 1, 2006	\$85,000.00								
Series 2005 Bonds									
Interest Due September 1, 2006	\$863,076.33								
Principal Due September 1, 2006	\$80,000.00								
Direct Construction Funding	\$2,941,081.31								
Projected Surplus/(Draw on Reserve)		\$0.00							
FY 2006-07 Obligations		\$5,256,325.64							
Series 2002 Bonds									
Interest Due March 1, 2007	\$670,982.50								
Interest Due September 1, 2007	\$670,982.50								
Principal Due September 1, 2007	\$115,000.00								
Series 2005 Bonds									
Interest Due March 1, 2007	\$1,112,446.88								
Interest Due September 1, 2007	\$1,112,446.88								
Principal Due September 1, 2007	\$115,000.00								
Administrative Expense Budget for Fiscal Year 2006-07	\$43,297.29								
Direct Construction Funding	\$1,416,169.59								
<u> </u>									
Special Tax Requirement for FY 2006-07		\$5,256,325.64							

IV. SPECIAL TAX CLASSIFICATIONS AND DEVELOPMENT UPDATE

CFD No. 6 encompasses approximately 2,888 gross acres of land located in the unincorporated portion of the County. At build-out, CFD No. 6 is expected to be developed into 4,715 Units.

A. Special Tax Classifications

Each Fiscal Year, the District calculates the Annual Special Taxes to be levied against Taxable Property within CFD No. 6. The RMA defines two (2) categories of Taxable Property, "Developed Property" and "Undeveloped Property." Developed Property is in turn divided into four (4) separate rate classifications which vary with Land Use Class (i.e., Residential Property and Commercial/Industrial Property) and Unit type (i.e., Detached Unit, Attached Unit and Affordable Unit). Developed Property is distinguished from Undeveloped Property by the issuance of a Building Permit. Specifically, Developed Property means, for any Fiscal Year, all Assessor's Parcels in CFD No. 6 for which a Building Permit for a Unit has been issued as of January 1 of the previous Fiscal Year. For example, all Assessor's Parcels in CFD No. 6 for which a Building Permit was issued as of January 1, 2006 will be classified as Developed Property in Fiscal Year 2006-07.

B. Development Update

As of January 1, 2006, the County had issued Building Permits for 2,575 Units, which equates to CFD No. 6 being 54.61 percent built-out in Fiscal Year 2006-07. Table 6 lists the Developed Property of CFD No. 6 by Special Tax Class for Fiscal Years 2004-05 through 2006-07.

TABLE 6
STATUS OF DEVELOPED PROPERTY

Unit Type	Fiscal Year 2004-05	Fiscal Year 2005-06	Fiscal Year 2006-07
Detached Unit	1,551	1,630	2,116
Attached Unit	0	0	339
Affordable Unit	120	120	120
Commercial/Industrial Property	NA	NA	NA
Total	1,671	1,750	2,575

V. METHOD OF APPORTIONMENT

Section F of the RMA describes the steps in which Annual Special Taxes shall be levied each Fiscal Year in order to satisfy the Annual Special Tax Requirement.

The first step states that the Annual Special Tax shall be levied against each parcel of Developed Property at the Assigned Annual Special Tax rate applicable to such Assessor's Parcel. If the Special Taxes raised pursuant to the first step are less than the Special Tax Requirement, then the second step is applied. This step states that the Annual Special Tax shall be levied Proportionately on each Assessor's Parcel of Undeveloped Property up to the Assigned Annual Special Tax rate applicable to such Assessor's Parcel. If the Annual Special Taxes raised pursuant to the first two steps are less than the Special Tax Requirement, then the third step is applied. This step states that the Annual Special Tax on each Assessor's Parcel of Developed Property whose Maximum Special Tax is the Backup Annual Special Tax shall be increased Proportionately from the Assigned Annual Special Tax rate up to the Backup Annual Special Tax rate in order to satisfy the Special Tax Requirement.

Application of the first step yields sufficient Special Taxes to satisfy the Special Tax Requirement for Fiscal Year 2006-07. The Initial Special Tax rates of CFD No. 6 for Fiscal Year 2006-07 are shown in Table 7. The Special Tax roll, which lists the actual Annual Special Tax levied against each Assessor's Parcel, is included as Exhibit B.

TABLE 7
INITIAL SPECIAL TAX RATES
FOR FISCAL YEAR 2006-07

Land Use Class	Unit Type	Initial Annual Special Tax
Residential Property	Detached Unit	\$2,462.36
Residential Property	Attached Unit	\$1,089.12
Residential Property	Affordable Unit	\$0.00
Commercial/Industrial Property	NA	\$0.00

J:\CLIENTS\POWAY.USD\CFD No. 6 - 4S Ranch\Admin\06-07\AdmReport_0607.doc

EXHIBIT A

SUMMARY OF TRANSACTIONS WITHIN FUNDS AND ACCOUNTS FISCAL YEAR 2005-06



Year to Date as of 4/30/2006

			SOURCES					USES					
Account Name	Beginning Balance	Special Tax Receipts	Investment Earnings	Miscellaneous	Total Sources	Interest Payments	Principal Payments	Acquisition/ Construction Payments	Legal/ Consulting/ Professional Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$0.00	\$0.00	\$973.42	\$247,000.00	\$247,973.42	\$0.00	\$0.00	\$0.00	(\$214,030.77)	\$0.00	(\$214,030.77)	\$0.00	\$33,942.65
2005 School Facilities Account 7150804K	\$0.00	\$0.00	\$424,281.94	\$39,057,437.37	\$39,481,719.31	\$0.00	\$0.00	(\$20,682,945.62)	\$0.00	\$0.00	(\$20,682,945.62)	\$0.00	\$18,798,773.69
4S Ranch Middle School Account 7150804N	\$1,511,524.35	\$0.00	\$5,541.54	\$0.00	\$5,541.54	\$0.00	\$0.00	(\$1,517,064.64)	\$0.00	\$0.00	(\$1,517,064.64)	\$0.00	\$1.25
Administrative Expense Fund 7150804I	\$65,416.97	\$0.00	\$1,436.01	\$27.00	\$1,463.01	\$0.00	\$0.00	\$0.00	(\$67,111.81)	\$0.00	(\$67,111.81)	\$42,448.32	\$42,216.49
Interest Account 7150804B	\$0.00	\$0.00	\$486.53	\$0.00	\$486.53	(\$1,952,640.12)	\$0.00	\$0.00	\$0.00	\$0.00	(\$1,952,640.12)	\$1,952,252.57	\$98.98
Prepayment Account 7150804M	\$0.00	\$0.00	\$50,540.91	\$4,813,915.50	\$4,864,456.41	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$4,864,456.41
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$55,000.00)	\$0.00	\$0.00	\$0.00	(\$55,000.00)	\$55,000.00	\$0.00
Reserve Fund 7150804D	\$2,312,214.80	\$0.00	\$133,830.57	\$3,901,513.25	\$4,035,343.82	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$32,067.00)	\$6,315,491.62
School Facilities Fund 7150804E	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$2,876,841.78	\$2,341,663.22	\$77,779.60	\$0.00	\$2,419,442.82	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$2,017,633.89)	\$3,278,650.71
Totals	\$6,765,997.90	\$2,341,663.22	\$694,870.52	\$48,019,893.12	\$51,056,426.86	(\$1,952,640.12)	(\$55,000.00)	(\$22,200,010.26	(\$281,142.58)	\$0.00	(\$24,488,792.96	\$0.00	\$33,333,631.80

 Prepared On:
 9/8/2006

Poway Unified School District CFD 6 Series 2002 & 2005

7/1/2005

			SOURCES					USES					
Account Name	Beginning Balance	Special Tax Receipts	Investment Earnings	Miscellaneous	Total Sources	Interest Payments	Principal Payments	Acquisition/ Construction Payments	Legal/ Consulting/ Professional Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
2005 School Facilities Account 7150804K	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
4S Ranch Middle School Account 7150804N	\$1,511,524.35	\$0.00	\$3,142.96	\$0.00	\$3,142.96	\$0.00	\$0.00	(\$1,504,000.00)	\$0.00	\$0.00	(\$1,504,000.00)	\$0.00	\$10,667.31
Administrative Expense Fund 7150804I	\$65,416.97	\$0.00	\$146.65	\$0.00	\$146.65	\$0.00	\$0.00	\$0.00	(\$4,802.38)	\$0.00	(\$4,802.38)	\$0.00	\$60,761.24
Interest Account 7150804B	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Prepayment Account 7150804M	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund 7150804D	\$2,312,214.80	\$0.00	\$23.91	\$0.00	\$23.91	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,312,238.71
School Facilities Fund 7150804E	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$2,876,841.78	\$34,980.06	\$5,923.90	\$0.00	\$40,903.96	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,917,745.74
Totals	\$6,765,997.90	\$34,980.06	\$9,237.42	\$0.00	\$44,217.48	\$0.00	\$0.00	(\$1,504,000.00)	(\$4,802.38)	\$0.00	(\$1,508,802.38)	\$0.00	\$5,301,413.00

Note:

Prepared On: 9/8/2006 Page 1 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

8/1/2005

			SOURCES					USES					
Account Name	Beginning	Special Tax	Investment			Interest	Principal	Acquisition/ Construction	Legal/ Consulting/ Professional				
Account Name	Balance	Receipts	Earnings	Miscellaneous	Total Sources	Payments	Payments	Payments	Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
2005 School Facilities Account 7150804K	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
4S Ranch Middle School Account 7150804N	\$10,667.31	\$0.00	\$2,390.24	\$0.00	\$2,390.24	\$0.00	\$0.00	(\$12,891.42)	\$0.00	\$0.00	(\$12,891.42)	\$0.00	\$166.13
Administrative Expense Fund 7150804I	\$60,761.24	\$0.00	\$148.76	\$27.00	\$175.76	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$60,937.00
Interest Account 7150804B	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Prepayment Account 7150804M	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund 7150804D	\$2,312,238.71	\$0.00	\$36,405.41	\$0.00	\$36,405.41	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,348,644.12
School Facilities Fund 7150804E	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$2,917,745.74	\$7,225.21	\$6,745.92	\$0.00	\$13,971.13	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,931,716.87
Totals	\$5,301,413.00	\$7,225.21	\$45,690.33	\$27.00	\$52,942.54	\$0.00	\$0.00	(\$12,891.42)	\$0.00	\$0.00	(\$12,891.42)	\$0.00	\$5,341,464.12

Note: \$27.00 received from DTA resulting from overpayment.

Prepared On: 9/8/2006 Page 2 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

9/1/2005

	SOURCES						USES						
Account Name	Beginning Balance	Special Tax Receipts	Investment Earnings	Miscellaneous	Total Sources	Interest Payments	Principal Payments	Acquisition/ Construction Payments	Legal/ Consulting/ Professional Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
2005 School Facilities Account 7150804K	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
4S Ranch Middle School Account 7150804N	\$166.13	\$0.00	\$7.64	\$0.00	\$7.64	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$173.77
Administrative Expense Fund 7150804I	\$60,937.00	\$0.00	\$153.17	\$0.00	\$153.17	\$0.00	\$0.00	\$0.00	(\$9,001.65)	\$0.00	(\$9,001.65)	\$0.00	\$52,088.52
Interest Account 7150804B	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$673,295.00)	\$0.00	\$0.00	\$0.00	\$0.00	(\$673,295.00)	\$705,362.00	\$32,067.00
Prepayment Account 7150804M	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$55,000.00)	\$0.00	\$0.00	\$0.00	(\$55,000.00)	\$55,000.00	\$0.00
Reserve Fund 7150804D	\$2,348,644.12	\$0.00	\$2,330.12	\$0.00	\$2,330.12	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$32,067.00)	\$2,318,907.24
School Facilities Fund 7150804E	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$2,931,716.87	\$7,061.00	\$7,358.57	\$0.00	\$14,419.57	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$728,295.00)	\$2,217,841.44
Totals	\$5,341,464.12	\$7,061.00	\$9,849.50	\$0.00	\$16,910.50	(\$673,295.00)	(\$55,000.00)	\$0.00	(\$9,001.65)	\$0.00	(\$737,296.65)	\$0.00	\$4,621,077.97

Note: \$673,295.00 was paid in Interest and \$55,000.00 was paid in Principal for September debt service obligation.

Prepared On: 9/8/2006 Page 3 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

10/1/2005

			SOURCES					USES					
Account Name	Beginning Balance	Special Tax Receipts	Investment Earnings	Miscellaneous	Total Sources	Interest Payments	Principal Payments	Acquisition/ Construction Payments	Legal/ Consulting/ Professional Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
2005 School Facilities Account 7150804K	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
4S Ranch Middle School Account 7150804N	\$173.77	\$0.00	\$0.43	\$0.00	\$0.43	\$0.00	\$0.00	(\$173.22)	\$0.00	\$0.00	(\$173.22)	\$0.00	\$0.98
Administrative Expense Fund 7150804I	\$52,088.52	\$0.00	\$131.07	\$0.00	\$131.07	\$0.00	\$0.00	\$0.00	(\$4,129.02)	\$0.00	(\$4,129.02)	\$0.00	\$48,090.57
Interest Account 7150804B	\$32,067.00	\$0.00	\$13.50	\$0.00	\$13.50	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$32,080.50
Prepayment Account 7150804M	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund 7150804D	\$2,318,907.24	\$0.00	\$2,937.59	\$0.00	\$2,937.59	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,321,844.83
School Facilities Fund 7150804E	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$2,217,841.44	\$9,901.11	\$5,537.80	\$0.00	\$15,438.91	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,233,280.35
Totals	\$4,621,077.97	\$9,901.11	\$8,620.39	\$0.00	\$18,521.50	\$0.00	\$0.00	(\$173.22)	(\$4,129.02)	\$0.00	(\$4,302.24)	\$0.00	\$4,635,297.23

Note:

Prepared On: 9/8/2006 Page 4 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

11/1/2005

			SOURCES					USES					
Account Name	Beginning Balance	Special Tax Receipts	Investment Earnings	Miscellaneous	Total Sources	Interest Payments	Principal Payments	Acquisition/ Construction Payments	Legal/ Consulting/ Professional Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$0.00	\$0.00	\$0.00	\$247,000.00	\$247,000.00	\$0.00	\$0.00	\$0.00	(\$75,190.63)	\$0.00	(\$75,190.63)	\$0.00	\$171,809.37
2005 School Facilities Account 7150804K	\$0.00	\$0.00	\$0.00	\$39,057,240.90	\$39,057,240.90	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$39,057,240.90
4S Ranch Middle School Account 7150804N	\$0.98	\$0.00	\$0.27	\$0.00	\$0.27	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$1.25
Administrative Expense Fund 7150804I	\$48,090.57	\$0.00	\$130.11	\$0.00	\$130.11	\$0.00	\$0.00	\$0.00	(\$6,531.59)	\$0.00	(\$6,531.59)	\$0.00	\$41,689.09
Interest Account 7150804B	\$32,080.50	\$0.00	\$88.76	\$0.00	\$88.76	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$32,169.26
Prepayment Account 7150804M	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund 7150804D	\$2,321,844.83	\$0.00	\$3,114.60	\$3,901,513.25	\$3,904,627.85	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$6,226,472.68
School Facilities Fund 7150804E	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$2,233,280.35	\$31,219.62	\$5,968.92	\$0.00	\$37,188.54	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,270,468.89
Totals	\$4,635,297.23	\$31,219.62	\$9,302.66	\$43,205,754.15	\$43,246,276.43	\$0.00	\$0.00	\$0.00	(\$81,722.22)	\$0.00	(\$81,722.22)	\$0.00	\$47,799,851.44

Note: All Miscellaneous Sources of Funds are Initial Deposits of Bond Proceeds.

Prepared On: 9/8/2006 Page 5 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

12/1/2005

			SOURCES					USES					
	Beginning	Special Tax	Investment			Interest	Principal	Acquisition/ Construction	Legal/ Consulting/ Professional				
Account Name	Balance	Receipts	Earnings	Miscellaneous	Total Sources	Payments	Payments	Payments	Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance	\$171,809.37	\$0.00	\$191.72	\$0.00	\$191.72	\$0.00	\$0.00	\$0.00	(\$109,984.19)	\$0.00	(\$109,984.19)	\$0.00	\$62,016.90
7150804L													
2005 School Facilities Account 7150804K	\$39,057,240.90	\$0.00	\$32,504.10	\$0.00	\$32,504.10	\$0.00	\$0.00	(\$9,780,897.74)	\$0.00	\$0.00	(\$9,780,897.74)	\$0.00	\$29,308,847.26
4S Ranch Middle School Account	\$1.25	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$1.25
7150804N	7.1-0	7 - 1 - 1	7000	70.00		4	40100	7.1.7	7	7000	*****		,,,,,,
Administrative Expense Fund 7150804I	\$41,689.09	\$0.00	\$117.03	\$0.00	\$117.03	\$0.00	\$0.00	\$0.00	(\$2,835.11)	\$0.00	(\$2,835.11)	\$0.00	\$38,971.01
Interest Account	\$32,169.26	\$0.00	\$88.93	\$0.00	\$88.93	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$32,258.19
7150804B				· ·									
Prepayment Account	\$0.00	\$0.00	\$0.00	\$4,813,915.50	\$4,813,915.50	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$4,813,915.50
7150804M													
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund	\$6,226,472.68	\$0.00	\$6,474.39	\$0.00	\$6,474.39	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$6,232,947.07
7150804D	\$0,220,172.00	ψο.σσ	ψο, 17 1.55	ψ0.00	ψο, 17 1.59	Ψ0.00	Ψ0.00	Ψ0.00	ψ0.00	ψ0.00	\$0.00	Ψ0.00	90,232,717.07
School Facilities Fund	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
7150804E Closed	1	700	700	00		+3.00	+5.00	Ţ3.00	70.00	75.00			700
Special Tax Fund	\$2,270,468.89	\$623,625.44	\$6,250.63	\$0.00	\$629,876.07	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$2,900,344.96
7150804A		ļ					1						
Totals	\$47,799,851.44	\$623,625.44	\$45,626.80	\$4,813,915.50	\$5,483,167.74	\$0.00	\$0.00	(\$9,780,897.74)	(\$112,819.30)	\$0.00	(\$9,893,717.04)	\$0.00	\$43,389,302.14

Note: In the Prepayment Account, \$4,813,915.50 was a deposit for the prepayment of Special Taxes for the Affordable Housing Complex.

Prepared On: 9/8/2006 Page 6 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

1/1/2006

			SOURCES					USES					
Account Name	Beginning	Special Tax	Investment			Interest	Principal	Acquisition/ Construction	Legal/ Consulting/ Professional Services				
Account Name	Balance	Receipts	Earnings	Miscellaneous	Total Sources	Payments	Payments	Payments	Scrvices	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance	\$62,016.90	\$0.00	\$198.92	\$0.00	\$198.92	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$62,215.82
7150804L							·	·			· ·	· ·	
2005 School Facilities Account	\$29,308,847.26	\$0.00	\$102,757.47	\$0.00	\$102,757.47	\$0.00	\$0.00	(\$2,705,637.03)	\$0.00	\$0.00	(\$2,705,637.03)	\$0.00	\$26,705,967.70
7150804K													
4S Ranch Middle School Account 7150804N	\$1.25	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$1.25
Administrative Expense Fund	\$38,971.01	\$0.00	\$119.49	\$0.00	\$119.49	\$0.00	\$0.00	\$0.00	(\$10,000.00)	\$0.00	(\$10,000.00)	\$0.00	\$29,090.50
7150804I	, , , , , , , , , , , , , , , , , , , ,					,	,	,	(, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,	(, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	, , , , ,	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Interest Account	\$32,258.19	\$0.00	\$94.02	\$0.00	\$94.02	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$32,352.21
7150804B													
Prepayment Account	\$4,813,915.50	\$0.00	\$4,005.75	\$0.00	\$4,005.75	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$4,817,921.25
7150804M													
Principal Account	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
7150804C													
Reserve Fund 7150804D	\$6,232,947.07	\$0.00	\$14,793.13	\$0.00	\$14,793.13	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$6,247,740.20
School Facilities Fund													
7150804E Closed	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund													
7150804A	\$2,900,344.96	\$1,188,038.73	\$7,577.67	\$0.00	\$1,195,616.40	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$4,095,961.36
Totals	\$43,389,302.14	\$1,188,038.73	\$129,546.45	\$0.00	\$1,317,585.18	\$0.00	\$0.00	(\$2,705,637.03)	(\$10,000.00)	\$0.00	(\$2,715,637.03)	\$0.00	\$41,991,250.29

Note:

Prepared On: 9/8/2006 Page 7 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

2/1/2006

			SOURCES					USES					
Account Name	Beginning	Special Tax	Investment		m. 10	Interest	Principal	Acquisition/ Construction	Legal/ Consulting/ Professional Services				
	Balance	Receipts	Earnings	Miscellaneous	Total Sources	Payments	Payments	Payments	Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$62,215.82	\$0.00	\$196.80	\$0.00	\$196.80	\$0.00	\$0.00	\$0.00	(\$3,225.00)	\$0.00	(\$3,225.00)	\$0.00	\$59,187.62
2005 School Facilities Account 7150804K	\$26,705,967.70	\$0.00	\$115,133.15	\$0.00	\$115,133.15	\$0.00	\$0.00	(\$1,784,979.12)	\$0.00	\$0.00	(\$1,784,979.12)	\$0.00	\$25,036,121.73
4S Ranch Middle School Account 7150804N	\$1.25	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$1.25
Administrative Expense Fund 7150804I	\$29,090.50	\$0.00	\$110.03	\$0.00	\$110.03	\$0.00	\$0.00	\$0.00	(\$10,719.75)	\$0.00	(\$10,719.75)	\$42,448.32	\$60,929.10
Interest Account 7150804B	\$32,352.21	\$0.00	\$102.34	\$0.00	\$102.34	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$32,454.55
Prepayment Account 7150804M	\$4,817,921.25	\$0.00	\$15,241.74	\$0.00	\$15,241.74	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$4,833,162.99
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund 7150804D	\$6,247,740.20	\$0.00	\$58,776.37	\$0.00	\$58,776.37	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$6,306,516.57
School Facilities Fund 7150804E Closed	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$4,095,961.36	\$45,587.81	\$9,960.19	\$0.00	\$55,548.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$42,448.32)	\$4,109,061.04
Totals	\$41,991,250.29	\$45,587.81	\$199,520.62	\$0.00	\$245,108.43	\$0.00	\$0.00	(\$1,784,979.12)	(\$13,944.75)	\$0.00	(\$1,798,923.87)	\$0.00	\$40,437,434.85

Note:

Prepared On: 9/8/2006 Page 8 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

3/1/2006

			SOURCES					USES					
						•		Acquisition/ Construction	Legal/ Consulting/ Professional				
Account Name	Beginning Balance	Special Tax Receipts	Investment Earnings	Miscellaneous	Total Sources	Interest Payments	Principal Payments	Payments	Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$59,187.62	\$0.00	\$185.46	\$0.00	\$185.46	\$0.00	\$0.00	\$0.00	(\$25,630.95)	\$0.00	(\$25,630.95)	\$0.00	\$33,742.13
2005 School Facilities Account 7150804K	\$25,036,121.73	\$0.00	\$103,231.60	\$196.47	\$103,428.07	\$0.00	\$0.00	(\$4,386,922.63)	\$0.00	\$0.00	(\$4,386,922.63)	\$0.00	\$20,752,627.17
4S Ranch Middle School Account 7150804N	\$1.25	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$1.25
Administrative Expense Fund 7150804I	\$60,929.10	\$0.00	\$170.43	\$0.00	\$170.43	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$61,099.53
Interest Account 7150804B	\$32,454.55	\$0.00	\$98.64	\$0.00	\$98.64	(\$1,279,345.12)	\$0.00	\$0.00	\$0.00	\$0.00	(\$1,279,345.12)	\$1,246,890.57	\$98.64
Prepayment Account 7150804M	\$4,833,162.99	\$0.00	\$14,688.99	\$0.00	\$14,688.99	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$4,847,851.98
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund 7150804D	\$6,306,516.57	\$0.00	\$1,229.25	\$0.00	\$1,229.25	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$6,307,745.82
School Facilities Fund 7150804E Closed	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$4,109,061.04	\$88,661.54	\$12,441.78	\$0.00	\$101,103.32	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	(\$1,246,890.57)	\$2,963,273.79
Totals	\$40,437,434.85	\$88,661.54	\$132,046.15	\$196.47	\$220,904.16	(\$1,279,345.12)	\$0.00	(\$4,386,922.63)	(\$25,630.95)	\$0.00	(\$5,691,898.70)	\$0.00	\$34,966,440.31

Note: In the School Facilities Account, Miscellaneous Use of Funds, there was an error in transfer in the amount of \$196.47 paid to SCC, LLC.

Prepared On: 9/8/2006 Page 9 of 10

Poway Unified School District CFD 6 Series 2002 & 2005

4/1/2006

		SOURCES				USES							
Account Name	Beginning	Special Tax	Investment			Interest	Principal	Acquisition/ Construction	Legal/ Consulting/ Professional				
Account Name	Balance	Receipts	Earnings	Miscellaneous	Total Sources	Payments	Payments	Payments	Services	Miscellaneous	Total Uses	Transfers	Ending Balance
2005 Cost of Issuance 7150804L	\$33,742.13	\$0.00	\$200.52	\$0.00	\$200.52	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$33,942.65
2005 School Facilities Account 7150804K	\$20,752,627.17	\$0.00	\$70,655.62	\$0.00	\$70,655.62	\$0.00	\$0.00	(\$2,024,509.10)	\$0.00	\$0.00	(\$2,024,509.10)	\$0.00	\$18,798,773.69
4S Ranch Middle School Account 7150804N	\$1.25	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$1.25
Administrative Expense Fund 7150804I	\$61,099.53	\$0.00	\$209.27	\$0.00	\$209.27	\$0.00	\$0.00	\$0.00	(\$19,092.31)	\$0.00	(\$19,092.31)	\$0.00	\$42,216.49
Interest Account 7150804B	\$98.64	\$0.00	\$0.34	\$0.00	\$0.34	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$98.98
Prepayment Account 7150804M	\$4,847,851.98	\$0.00	\$16,604.43	\$0.00	\$16,604.43	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$4,864,456.41
Principal Account 7150804C	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Reserve Fund 7150804D	\$6,307,745.82	\$0.00	\$7,745.80	\$0.00	\$7,745.80	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$6,315,491.62
School Facilities Fund 7150804E Closed	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Special Tax Fund 7150804A	\$2,963,273.79	\$305,362.70	\$10,014.22	\$0.00	\$315,376.92	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$3,278,650.71
Totals	\$34,966,440.31	\$305,362.70	\$105,430.20	\$0.00	\$410,792.90	\$0.00	\$0.00	(\$2,024,509.10)	(\$19,092.31)	\$0.00	(\$2,043,601.41)	\$0.00	\$33,333,631.80

Note:

Prepared On: 9/8/2006 Page 10 of 10

EXHIBIT B

ANNUAL SPECIAL TAX RATES FISCAL YEAR 2006-07

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	520	1	00	\$2,240.94
678	520	2	00	\$2,240.94
678	520	3	00	\$2,240.94
678	520	4	00	\$2,240.94
678	520	5	00	\$2,240.94
678	520	6	00	\$2,240.94
678	520	7	00	\$2,240.94
678	520	8	00	\$2,240.94
678	520	9	00	\$2,240.94
678	520	10	00	\$2,240.94
678	520	11	00	\$2,240.94
678	520	12	00	\$2,240.94
678	520	13	00	\$2,240.94
678	520	14	00	\$2,240.94
678	520	15	00	\$2,240.94
678	520	16	00	\$2,240.94
678	520	17	00	\$2,240.94
678	520	18	00	\$2,240.94
678	520	19	00	\$2,240.94
678	520	20	00	\$2,240.94
678	520	21	00	\$2,240.94
678	520	22	00	\$2,240.94
678	520	23	00	\$2,240.94
678	520	59	00	\$2,240.94
678	520	60	00	\$2,240.94
678	520	26	00	\$2,240.94
678	520	27	00	\$2,240.94
678	520	28	00	\$2,240.94
678	520	29	00	\$2,240.94
678	520	30	00	\$2,240.94
678	520	31	00	\$2,240.94
678	520	32	00	\$2,240.94
678	520	33	00	\$2,240.94
678	520	34	00	\$2,240.94
678	520	35	00	\$2,240.94
678	520	36	00	\$2,240.94
678	520	37	00	\$2,240.94
678	520	38	00	\$2,240.94
678	520	39	00	\$2,240.94
678	520	40	00	\$2,240.94
678	520	41	00	\$2,240.94
678	520	42	00	\$2,240.94

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	520	43	00	\$2,240.94
678	520	44	00	\$2,240.94
678	520	45	00	\$2,240.94
678	520	46	00	\$2,240.94
678	520	47	00	\$2,240.94
678	520	48	00	\$2,240.94
678	520	49	00	\$2,240.94
678	520	50	00	\$2,240.94
678	520	51	00	\$2,240.94
678	520	52	00	\$2,240.94
678	520	53	00	\$2,240.94
678	520	54	00	\$2,240.94
678	520	55	00	\$2,240.94
678	520	56	00	\$2,240.94
678	520	57	00	\$2,240.94
678	520	58	00	\$0.00
678	521	1	00	\$2,240.94
678	521	2	00	\$2,240.94
678	521	3	00	\$2,240.94
678	521	4	00	\$2,240.94
678	521	5	00	\$2,240.94
678	521	6	00	\$2,240.94
678	521	7	00	\$2,240.94
678	521	8	00	\$2,240.94
678	521	9	00	\$2,240.94
678	521	20	00	\$2,240.94
678	521	21	00	\$2,240.94
678	521	12	00	\$2,240.94
678	521	13	00	\$2,240.94
678	521	14	00	\$2,240.94
678	521	15	00	\$2,240.94
678	521	16	00	\$2,240.94
678	521	17	00	\$2,240.94
678	521	18	00	\$2,240.94
678	521	19	00	\$0.00
678	540	1	00	\$0.00
678	540	2	00	\$2,240.94
678	540	3	00	\$2,240.94
678	540	4	00	\$2,240.94
678	540	5	00	\$2,240.94
678	540	6	00	\$2,240.94
678	540	7	00	\$2,240.94

	ASSESSOR PARCEL NUMBER								
воок	PAGE	PARCEL	INT	TAX					
678	540	8	00	\$2,240.94					
678	540	9	00	\$2,240.94					
678	540	10	00	\$2,240.94					
678	540	11	00	\$2,240.94					
678	540	12	00	\$2,240.94					
678	540	13	00	\$2,240.94					
678	540	14	00	\$2,240.94					
678	540	15	00	\$2,240.94					
678	540	16	00	\$2,240.94					
678	540	17	00	\$2,240.94					
678	540	18	00	\$2,240.94					
678	540	19	00	\$2,240.94					
678	540	20	00	\$2,240.94					
678	540	21	00	\$2,240.94					
678	540	22	00	\$2,240.94					
678	540	23	00	\$2,240.94					
678	540	24	00	\$2,240.94					
678	540	25	00	\$2,240.94					
678	540	26	00	\$2,240.94					
678	540	27	00	\$2,240.94					
678	540	28	00	\$2,240.94					
678	540	29	00	\$2,240.94					
678	540	30	00	\$2,240.94					
678	540	31	00	\$2,240.94					
678	540	32	00	\$2,240.94					
678	540	33	00	\$2,240.94					
678	540	34	00	\$2,240.94					
678	540	35	00	\$2,240.94					
678	540	36	00	\$2,240.94					
678	540	37	00	\$2,240.94					
678	540	38	00	\$2,240.94					
678	540	39	00	\$2,240.94					
678	540	40	00	\$2,240.94					
678	540	41	00	\$2,240.94					
678	541	1	00	\$2,240.94					
678	541	2	00	\$2,240.94					
678	541	3	00	\$2,240.94					
678	541	4	00	\$2,240.94					
678	541	5	00	\$2,240.94					
678	541	6	00	\$2,240.94					
678	541	7	00	\$2,240.94					
678	541	8	00	\$2,240.94					

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	541	9	00	\$2,240.94
678	541	10	00	\$2,240.94
678	541	11	00	\$2,240.94
678	541	12	00	\$2,240.94
678	541	13	00	\$2,240.94
678	541	14	00	\$2,240.94
678	541	15	00	\$2,240.94
678	541	16	00	\$2,240.94
678	541	17	00	\$2,240.94
678	541	18	00	\$2,240.94
678	541	19	00	\$2,240.94
678	541	20	00	\$2,240.94
678	541	21	00	\$2,240.94
678	541	22	00	\$2,240.94
678	541	23	00	\$2,240.94
678	541	24	00	\$2,240.94
678	541	25	00	\$2,240.94
678	541	26	00	\$2,240.94
678	541	27	00	\$2,240.94
678	541	28	00	\$2,240.94
678	541	29	00	\$2,240.94
678	541	30	00	\$2,240.94
678	541	31	00	\$2,240.94
678	541	32	00	\$2,240.94
678	541	33	00	\$2,240.94
678	541	34	00	\$2,240.94
678	541	35	00	\$2,240.94
678	541	36	00	\$2,240.94
678	541	37	00	\$2,240.94
678	541	38	00	\$2,240.94
678	541	39	00	\$2,240.94
678	541	40	00	\$2,240.94
678	541	41	00	\$0.00
678	541	42	00	\$0.00
678	530	1	00	\$2,253.68
678	530	2	00	\$2,253.68
678	530	3	00	\$2,253.68
678	530	4	00	\$2,253.68
678	530	5	00	\$2,253.68
678	530	6	00	\$2,253.68
678	530	7	00	\$2,253.68
678	530	8	00	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	530	9	00	\$2,253.68
678	530	10	00	\$2,253.68
678	530	11	00	\$2,253.68
678	530	12	00	\$2,253.68
678	530	13	00	\$2,253.68
678	530	14	00	\$2,253.68
678	530	15	00	\$2,253.68
678	530	16	00	\$2,253.68
678	530	41	00	\$2,253.68
678	530	18	00	\$2,253.68
678	530	19	00	\$2,253.68
678	530	20	00	\$2,253.68
678	530	21	00	\$2,253.68
678	530	22	00	\$2,253.68
678	530	23	00	\$2,253.68
678	530	24	00	\$2,253.68
678	530	25	00	\$2,253.68
678	530	26	00	\$2,253.68
678	530	27	00	\$2,253.68
678	530	28	00	\$2,253.68
678	530	29	00	\$2,253.68
678	530	30	00	\$2,253.68
678	530	31	00	\$2,253.68
678	530	32	00	\$2,253.68
678	530	33	00	\$2,253.68
678	530	34	00	\$2,253.68
678	530	35	00	\$2,240.94
678	530	36	00	\$2,240.94
678	530	37	00	\$2,240.94
678	530	38	00	\$2,253.68
678	530	39	00	\$0.00
678	530	40	00	\$0.00
678	531	1	00	\$2,253.68
678	531	75	00	\$2,253.68
678	531	76	00	\$2,253.68
678	531	4	00	\$2,253.68
678	531	68	00	\$2,253.68
678	531	69	00	\$2,253.68
678	531	70	00	\$2,253.68
678	531	71	00	\$2,253.68
678	531	72	00	\$2,240.94
678	531	10	00	\$2,240.94

	SPECIAL			
воок	TAX			
678	531	11	00	\$2,240.94
678	531	12	00	\$2,240.94
678	531	13	00	\$2,240.94
678	531	14	00	\$2,240.94
678	531	15	00	\$2,240.94
678	531	16	00	\$2,240.94
678	531	17	00	\$2,253.68
678	531	18	00	\$2,253.68
678	531	19	00	\$2,253.68
678	531	20	00	\$2,253.68
678	531	21	00	\$2,253.68
678	531	22	00	\$2,253.68
678	531	73	00	\$2,253.68
678	531	24	00	\$2,253.68
678	531	74	00	\$2,253.68
678	531	26	00	\$2,253.68
678	531	27	00	\$2,253.68
678	531	28	00	\$2,253.68
678	531	29	00	\$2,253.68
678	531	30	00	\$2,253.68
678	531	31	00	\$2,253.68
678	531	32	00	\$2,240.94
678	531	33	00	\$2,240.94
678	531	34	00	\$2,240.94
678	531	35	00	\$2,240.94
678	531	36	00	\$2,240.94
678	531	37	00	\$2,240.94
678	531	38	00	\$2,240.94
678	531	39	00	\$2,240.94
678	531	40	00	\$2,240.94
678	531	41	00	\$2,240.94
678	531	42	00	\$2,240.94
678	531	43	00	\$2,240.94
678	531	44	00	\$2,240.94
678	531	45	00	\$2,240.94
678	531	46	00	\$2,240.94
678	531	47	00	\$2,240.94
678	531	48	00	\$2,240.94
678	531	49	00	\$2,240.94
678	531	50	00	\$2,240.94
678	531	51	00	\$2,240.94
678	531	52	00	\$2,240.94

	SPECIAL			
BOOK PAGE PARCEL INT				TAX
678	531	53	00	\$2,240.94
678	531	54	00	\$2,240.94
678	531	55	00	\$2,240.94
678	531	56	00	\$2,240.94
678	531	57	00	\$2,240.94
678	531	58	00	\$2,240.94
678	531	59	00	\$2,240.94
678	531	60	00	\$2,240.94
678	531	61	00	\$2,240.94
678	531	62	00	\$2,240.94
678	531	63	00	\$2,240.94
678	531	64	00	\$2,240.94
678	531	65	00	\$2,240.94
678	531	66	00	\$0.00
678	531	67	00	\$0.00
678	550	1	00	\$2,253.68
678	550	2	00	\$2,240.94
678	550	3	00	\$2,240.94
678	550	4	00	\$2,240.94
678	550	5	00	\$2,240.94
678	550	6	00	\$2,253.68
678	550	7	00	\$2,253.68
678	550	8	00	\$2,240.94
678	550	9	00	\$2,240.94
678	550	10	00	\$2,240.94
678	550	11	00	\$2,240.94
678	550	12	00	\$2,240.94
678	550	13	00	\$2,240.94
678	550	14	00	\$2,240.94
678	550	15	00	\$2,240.94
678	550	16	00	\$2,240.94
678	550	17	00	\$2,240.94
678	550	18	00	\$2,240.94
678	550	37	00	\$2,240.94
678	550	39	00	\$0.00
678	550	40	00	\$2,240.94
678	550	36	00	\$2,240.94
678	550	22	00	\$2,240.94
678	550	23	00	\$2,240.94
678	550	24	00	\$2,240.94
678	550	25	00	\$2,253.68
678	550	26	00	\$2,253.68

	SPECIAL			
BOOK PAGE PARCEL INT				TAX
678	550	27	00	\$2,253.68
678	550	28	00	\$2,240.94
678	550	29	00	\$2,253.68
678	550	30	00	\$2,253.68
678	550	31	00	\$0.00
678	550	32	00	\$0.00
678	551	1	00	\$2,240.94
678	551	2	00	\$2,240.94
678	551	3	00	\$2,240.94
678	551	4	00	\$2,240.94
678	551	5	00	\$2,240.94
678	551	6	00	\$2,240.94
678	551	7	00	\$2,240.94
678	551	8	00	\$2,240.94
678	551	9	00	\$2,240.94
678	551	10	00	\$2,240.94
678	551	11	00	\$2,253.68
678	551	12	00	\$2,253.68
678	551	13	00	\$2,253.68
678	551	14	00	\$2,253.68
678	551	15	00	\$2,253.68
678	551	16	00	\$2,253.68
678	551	17	00	\$2,253.68
678	551	18	00	\$2,253.68
678	551	19	00	\$2,253.68
678	551	20	00	\$2,240.94
678	551	21	00	\$2,240.94
678	551	22	00	\$2,240.94
678	551	23	00	\$2,240.94
678	551	24	00	\$2,240.94
678	551	25	00	\$2,240.94
678	551	26	00	\$2,240.94
678	551	27	00	\$2,240.94
678	552	1	00	\$2,240.94
678	552	2	00	\$2,240.94
678	552	3	00	\$2,253.68
678	552	4	00	\$2,253.68
678	552	5	00	\$2,253.68
678	552	6	00	\$2,253.68
678	552	7	00	\$2,253.68
678	552	8	00	\$2,253.68
678	552	9	00	\$2,253.68

	SPECIAL			
воок	TAX			
678	552	10	00	\$2,253.68
678	552	11	00	\$2,253.68
678	552	12	00	\$2,253.68
678	552	13	00	\$2,253.68
678	552	14	00	\$2,253.68
678	552	15	00	\$2,253.68
678	552	16	00	\$2,253.68
678	552	17	00	\$2,253.68
678	552	18	00	\$2,253.68
678	552	19	00	\$2,253.68
678	552	20	00	\$2,253.68
678	552	21	00	\$2,253.68
678	552	22	00	\$2,253.68
678	552	23	00	\$2,253.68
678	552	24	00	\$2,253.68
678	552	25	00	\$2,253.68
678	552	26	00	\$2,253.68
678	552	27	00	\$2,253.68
678	552	28	00	\$2,240.94
678	552	29	00	\$2,253.68
678	552	30	00	\$2,253.68
678	552	31	00	\$2,253.68
678	552	32	00	\$2,253.68
678	552	33	00	\$2,253.68
678	552	34	00	\$2,240.94
678	553	1	00	\$2,240.94
678	553	2	00	\$2,240.94
678	553	3	00	\$2,240.94
678	553	4	00	\$2,240.94
678	553	5	00	\$2,240.94
678	553	6	00	\$2,240.94
678	553	7	00	\$2,240.94
678	553	8	00	\$2,240.94
678	553	9	00	\$2,240.94
678	553	10	00	\$2,240.94
678	553	11	00	\$2,240.94
678	553	12	00	\$2,240.94
678	553	13	00	\$2,240.94
678	553	14	00	\$2,240.94
678	553	15	00	\$2,240.94
678	553	16	00	\$2,240.94
678	553	17	00	\$2,240.94

	SPECIAL			
BOOK PAGE PARCEL INT				TAX
678	553	18	00	\$2,240.94
678	553	19	00	\$2,240.94
678	553	20	00	\$2,240.94
678	553	21	00	\$2,240.94
678	553	22	00	\$2,240.94
678	553	23	00	\$2,240.94
678	553	24	00	\$2,240.94
678	553	25	00	\$2,240.94
678	553	26	00	\$2,240.94
678	553	27	00	\$2,240.94
678	553	28	00	\$2,240.94
678	553	29	00	\$2,240.94
678	553	30	00	\$2,240.94
678	553	31	00	\$2,240.94
678	553	32	00	\$2,240.94
678	553	33	00	\$2,240.94
678	553	34	00	\$2,240.94
678	553	35	00	\$2,240.94
678	560	1	00	\$2,253.68
678	560	2	00	\$2,240.94
678	560	3	00	\$2,240.94
678	560	4	00	\$2,240.94
678	560	5	00	\$2,240.94
678	560	6	00	\$2,253.68
678	560	7	00	\$2,253.68
678	560	8	00	\$2,253.68
678	560	9	00	\$2,253.68
678	560	10	00	\$2,253.68
678	560	11	00	\$2,253.68
678	560	12	00	\$2,253.68
678	560	13	00	\$2,253.68
678	560	14	00	\$2,253.68
678	560	15	00	\$2,253.68
678	560	16	00	\$2,253.68
678	560	17	00	\$2,253.68
678	560	18	00	\$2,240.94
678	560	19	00	\$2,240.94
678	560	20	00	\$2,240.94
678	560	21	00	\$2,240.94
678	560	22	00	\$2,240.94
678	560	23	00	\$2,240.94
678	560	24	00	\$2,240.94

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	560	25	00	\$2,240.94	
678	560	26	00	\$2,240.94	
678	560	27	00	\$2,240.94	
678	560	28	00	\$2,240.94	
678	560	29	00	\$2,240.94	
678	560	30	00	\$2,240.94	
678	560	31	00	\$2,240.94	
678	560	32	00	\$2,240.94	
678	560	33	00	\$2,240.94	
678	560	34	00	\$0.00	
678	560	35	00	\$0.00	
678	561	1	00	\$2,240.94	
678	561	2	00	\$2,240.94	
678	561	3	00	\$2,240.94	
678	561	4	00	\$2,253.68	
678	561	5	00	\$2,253.68	
678	561	6	00	\$2,253.68	
678	561	7	00	\$2,253.68	
678	561	8	00	\$2,253.68	
678	561	9	00	\$2,253.68	
678	561	10	00	\$2,253.68	
678	561	11	00	\$2,253.68	
678	561	12	00	\$2,253.68	
678	561	13	00	\$2,253.68	
678	561	14	00	\$2,253.68	
678	561	15	00	\$2,253.68	
678	561	16	00	\$2,253.68	
678	561	17	00	\$2,253.68	
678	561	18	00	\$2,253.68	
678	561	19	00	\$2,253.68	
678	561	20	00	\$2,240.94	
		21	00	1	
678	561			\$2,240.94	
678	561 561	22 23	00	\$2,240.94	
678 678				\$2,253.68	
	561	24	00	\$2,253.68	
678	561	25	00	\$2,253.68	
678	561	26	00	\$2,253.68	
678	561	27	00	\$2,253.68	
678	561	28	00	\$2,253.68	
678	561	29	00	\$2,253.68	
678	561	49	00	\$2,253.68	
678	561	50	00	\$2,253.68	

	SPECIAL			
BOOK PAGE PARCEL INT				TAX
678	561	32	00	\$0.00
678	561	33	00	\$2,253.68
678	561	34	00	\$2,253.68
678	561	35	00	\$2,253.68
678	561	36	00	\$2,253.68
678	561	37	00	\$2,240.94
678	561	38	00	\$2,240.94
678	561	39	00	\$2,240.94
678	561	40	00	\$2,240.94
678	561	41	00	\$2,240.94
678	561	42	00	\$2,240.94
678	561	43	00	\$2,240.94
678	561	44	00	\$2,240.94
678	561	45	00	\$2,240.94
678	561	46	00	\$0.00
678	561	47	00	\$0.00
678	561	48	00	\$0.00
678	562	1	00	\$2,253.68
678	562	2	00	\$2,253.68
678	562	3	00	\$2,253.68
678	562	4	00	\$2,253.68
678	562	5	00	\$2,253.68
678	562	6	00	\$2,253.68
678	562	7	00	\$2,253.68
678	562	8	00	\$2,253.68
678	562	9	00	\$2,253.68
678	562	10	00	\$2,253.68
678	562	11	00	\$2,253.68
678	562	12	00	\$2,253.68
678	562	13	00	\$2,253.68
678	562	14	00	\$2,253.68
678	562	15	00	\$2,253.68
678	562	16	00	\$2,253.68
678	562	17	00	\$2,253.68
678	562	18	00	\$2,253.68
678	562	19	00	\$2,253.68
678	562	20	00	\$2,253.68
678	562	21	00	\$2,253.68
678	562	22	00	\$2,253.68
678	562	23	00	\$2,253.68
678	562	24	00	\$2,253.68
678	562	25	00	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	562	26	00	\$2,253.68
678	562	27	00	\$2,253.68
678	562	28	00	\$2,253.68
678	562	29	00	\$2,253.68
678	562	30	00	\$2,253.68
678	562	31	00	\$2,253.68
678	562	32	00	\$2,240.94
678	562	33	00	\$2,240.94
678	562	34	00	\$2,240.94
678	562	35	00	\$2,240.94
678	562	36	00	\$2,240.94
678	562	37	00	\$2,240.94
678	562	38	00	\$2,240.94
678	562	39	00	\$2,240.94
678	562	40	00	\$2,240.94
678	562	41	00	\$2,240.94
678	562	42	00	\$2,240.94
678	562	43	00	\$2,240.94
678	562	44	00	\$2,240.94
678	562	45	00	\$2,240.94
678	570	1	00	\$2,253.68
678	570	2	00	\$2,253.68
678	570	3	00	\$2,253.68
678	570	4	00	\$2,253.68
678	570	5	00	\$2,253.68
678	570	6	00	\$2,253.68
678	570	7	00	\$2,253.68
678	570	8	00	\$2,253.68
678	570	9	00	\$2,253.68
678	570	10	00	\$2,253.68
678	570	11	00	\$2,253.68
678	570	12	00	\$2,253.68
678	570	13	00	\$2,253.68
678	570	14	00	\$2,253.68
678	570	15	00	\$2,253.68
678	570	16	00	\$2,253.68
678	570	17	00	\$2,253.68
678	570	18	00	\$2,253.68
678	570	19	00	\$2,253.68
678	570	20	00	\$2,253.68
678	570	21	00	\$2,253.68
678	570	22	00	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	570	23	00	\$2,253.68
678	570	24	00	\$2,253.68
678	570	25	00	\$2,253.68
678	570	26	00	\$2,253.68
678	570	27	00	\$2,253.68
678	570	28	00	\$2,253.68
678	570	29	00	\$0.00
678	570	30	00	\$0.00
678	571	1	00	\$2,240.94
678	571	2	00	\$2,240.94
678	571	3	00	\$2,240.94
678	571	4	00	\$2,240.94
678	571	5	00	\$2,240.94
678	571	6	00	\$2,240.94
678	571	7	00	\$2,240.94
678	571	8	00	\$2,240.94
678	571	9	00	\$2,240.94
678	571	10	00	\$2,240.94
678	571	11	00	\$2,240.94
678	571	12	00	\$2,240.94
678	571	13	00	\$2,240.94
678	571	14	00	\$2,240.94
678	571	15	00	\$2,253.68
678	571	16	00	\$2,253.68
678	571	17	00	\$2,253.68
678	571	18	00	\$2,253.68
678	571	19	00	\$2,253.68
678	571	20	00	\$2,253.68
678	571	21	00	\$2,253.68
678	571	22	00	\$2,253.68
678	571	23	00	\$2,253.68
678	571	24	00	\$2,253.68
678	571	25	00	\$2,253.68
678	571	26	00	\$2,253.68
678	571	27	00	\$2,253.68
678	571	28	00	\$2,253.68
678	571	29	00	\$2,253.68
678	571	30	00	\$2,253.68
678	571	31	00	\$2,253.68
678	571	33	00	\$0.00
678	572	1	00	\$2,253.68
678	572	2	00	\$2,253.68

	SPECIAL			
воок	PAGE	PARCEL	INT	TAX
678	572	3	00	\$2,253.68
678	572	4	00	\$2,253.68
678	572	5	00	\$2,253.68
678	572	6	00	\$2,253.68
678	572	7	00	\$2,253.68
678	572	8	00	\$2,253.68
678	572	9	00	\$2,253.68
678	572	10	00	\$2,253.68
678	572	11	00	\$2,253.68
678	572	12	00	\$2,253.68
678	572	13	00	\$2,253.68
678	572	14	00	\$2,253.68
678	572	15	00	\$2,253.68
678	572	16	00	\$2,253.68
678	572	17	00	\$2,253.68
678	572	18	00	\$2,253.68
678	572	19	00	\$2,253.68
678	572	20	00	\$2,253.68
678	572	21	00	\$2,253.68
678	572	22	00	\$2,253.68
678	572	23	00	\$2,253.68
678	572	24	00	\$2,253.68
678	572	25	00	\$2,253.68
678	572	26	00	\$2,253.68
678	572	27	00	\$2,265.82
678	572	28	00	\$2,265.82
678	572	29	00	\$2,265.82
678	572	30	00	\$2,265.82
678	572	31	00	\$2,265.82
678	572	32	00	\$2,265.82
678	572	33	00	\$2,265.82
678	572	34	00	\$2,265.82
678	572	35	00	\$2,265.82
678	572	36	00	\$2,265.82
678	572	37	00	\$2,265.82
678	572	38	00	\$2,265.82
678	572	39	00	\$2,265.82
678	572	40	00	\$2,265.82
678	572	41	00	\$2,265.82
678	572	42	00	\$2,265.82
678	572	43	00	\$2,265.82
678	572	44	00	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	572	45	00	\$2,240.94
678	572	46	00	\$2,240.94
678	572	47	00	\$2,240.94
678	572	48	00	\$2,265.82
678	572	49	00	\$2,265.82
678	572	50	00	\$0.00
678	572	51	00	\$0.00
678	580	1	00	\$2,240.94
678	580	2	00	\$2,240.94
678	580	3	00	\$2,240.94
678	580	4	00	\$2,240.94
678	580	5	00	\$2,240.94
678	580	6	00	\$2,240.94
678	580	7	00	\$2,240.94
678	580	8	00	\$2,240.94
678	580	9	00	\$2,240.94
678	580	10	00	\$2,240.94
678	580	11	00	\$2,240.94
678	580	12	00	\$2,240.94
678	580	13	00	\$2,240.94
678	580	14	00	\$2,240.94
678	580	15	00	\$2,240.94
678	580	16	00	\$2,240.94
678	580	17	00	\$2,240.94
678	580	18	00	\$2,240.94
678	580	19	00	\$2,240.94
678	580	20	00	\$2,240.94
678	580	21	00	\$2,240.94
678	580	22	00	\$2,240.94
678	580	23	00	\$2,240.94
678	580	24	00	\$2,253.68
678	580	25	00	\$2,253.68
678	580	26	00	\$2,253.68
678	580	27	00	\$2,253.68
678	580	28	00	\$2,253.68
678	580	29	00	\$2,253.68
678	580	30	00	\$2,253.68
678	580	31	00	\$2,253.68
678	580	32	00	\$2,253.68
678	580	33	00	\$2,253.68
678	580	34	00	\$2,253.68
678	580	35	00	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	580	36	00	\$2,253.68
678	580	37	00	\$2,253.68
678	580	38	00	\$2,253.68
678	580	39	00	\$2,253.68
678	580	40	00	\$2,240.94
678	580	41	00	\$2,240.94
678	580	42	00	\$2,240.94
678	580	43	00	\$2,253.68
678	580	44	00	\$2,253.68
678	580	45	00	\$2,253.68
678	580	46	00	\$2,253.68
678	580	47	00	\$0.00
678	580	48	00	\$0.00
678	580	49	00	\$0.00
678	590	1	00	\$2,240.94
678	590	2	00	\$2,240.94
678	590	3	00	\$2,240.94
678	590	4	00	\$2,240.94
678	590	5	00	\$2,240.94
678	590	6	00	\$2,240.94
678	590	7	00	\$2,240.94
678	590	8	00	\$2,240.94
678	590	9	00	\$2,240.94
678	590	10	00	\$2,253.68
678	590	11	00	\$2,253.68
678	590	12	00	\$2,253.68
678	590	13	00	\$2,253.68
678	590	14	00	\$2,253.68
678	590	15	00	\$2,253.68
678	590	16	00	\$2,253.68
678	590	17	00	\$2,253.68
678	590	18	00	\$2,253.68
678	590	19	00	\$2,253.68
678	590	20	00	\$2,253.68
678	590	21	00	\$2,253.68
678	590	22	00	\$2,253.68
678	590	23	00	\$2,253.68
678	590	24	00	\$2,253.68
678	590	25	00	\$2,253.68
678	590	26	00	\$2,240.94
678	590	27	00	\$2,240.94
678	590	28	00	\$2,240.94

	SPECIAL			
воок	PAGE	PARCEL	INT	TAX
678	590	29	00	\$2,240.94
678	590	30	00	\$2,240.94
678	590	31	00	\$2,240.94
678	590	32	00	\$2,240.94
678	590	33	00	\$2,240.94
678	590	34	00	\$2,240.94
678	590	35	00	\$0.00
678	590	36	00	\$0.00
312	260	1	00	\$2,265.82
312	260	2	00	\$2,265.82
312	260	3	00	\$2,265.82
312	260	4	00	\$2,265.82
312	260	5	00	\$2,265.82
312	260	6	00	\$2,265.82
312	260	7	00	\$2,265.82
312	260	8	00	\$2,265.82
312	260	9	00	\$2,265.82
312	260	10	00	\$2,265.82
312	260	11	00	\$2,265.82
312	260	12	00	\$2,265.82
312	260	13	00	\$2,265.82
312	260	14	00	\$2,265.82
312	260	15	00	\$2,265.82
312	260	16	00	\$2,265.82
312	260	17	00	\$2,265.82
312	260	18	00	\$2,265.82
312	260	19	00	\$2,265.82
312	260	20	00	\$2,265.82
312	260	21	00	\$2,265.82
312	260	22	00	\$2,265.82
312	260	23	00	\$2,265.82
312	260	24	00	\$2,265.82
312	260	25	00	\$2,265.82
312	260	26	00	\$2,265.82
312	260	27	00	\$2,265.82
312	260	28	00	\$2,265.82
312	260	29	00	\$2,265.82
312	260	30	00	\$2,265.82
312	260	31	00	\$2,265.82
312	260	32	00	\$2,253.68
312	260	33	00	\$2,253.68
312	260	34	00	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	260	35	00	\$2,265.82
312	260	36	00	\$2,265.82
312	260	37	00	\$2,265.82
312	260	38	00	\$2,265.82
312	260	39	00	\$2,265.82
312	260	40	00	\$2,265.82
312	260	41	00	\$2,265.82
312	260	42	00	\$2,265.82
312	260	43	00	\$2,265.82
312	260	44	00	\$2,265.82
312	260	45	00	\$2,265.82
312	260	46	00	\$2,265.82
312	260	47	00	\$2,265.82
312	260	48	00	\$2,265.82
312	260	49	00	\$2,265.82
312	261	1	00	\$2,265.82
312	261	2	00	\$2,265.82
312	261	3	00	\$2,265.82
312	261	4	00	\$2,265.82
312	261	5	00	\$2,265.82
312	261	6	00	\$2,253.68
312	261	7	00	\$2,253.68
312	261	8	00	\$2,253.68
312	261	9	00	\$2,253.68
312	261	10	00	\$2,253.68
312	261	11	00	\$2,253.68
312	261	12	00	\$2,253.68
312	261	13	00	\$2,253.68
312	261	14	00	\$2,253.68
312	261	15	00	\$2,253.68
312	261	16	00	\$2,253.68
312	261	17	00	\$2,253.68
312	261	18	00	\$2,253.68
312	261	19	00	\$2,253.68
312	261	20	00	\$2,253.68
312	261	21	00	\$2,253.68
312	261	22	00	\$2,253.68
312	261	23	00	\$2,253.68
312	261	24	00	\$2,253.68
312	261	25	00	\$2,253.68
312	261	26	00	\$2,265.82
312	261	27	00	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	261	28	00	\$2,265.82
312	261	29	00	\$2,265.82
312	261	30	00	\$2,265.82
312	261	31	00	\$2,265.82
312	262	1	00	\$2,265.82
312	262	2	00	\$2,265.82
312	262	3	00	\$2,265.82
312	262	4	00	\$2,265.82
312	262	5	00	\$2,265.82
312	262	6	00	\$2,265.82
312	262	7	00	\$2,265.82
312	262	8	00	\$2,265.82
312	262	9	00	\$2,265.82
312	262	10	00	\$2,265.82
312	262	11	00	\$2,265.82
312	262	12	00	\$2,265.82
312	262	13	00	\$2,265.82
312	262	14	00	\$2,265.82
312	262	15	00	\$2,265.82
312	262	16	00	\$2,265.82
312	262	17	00	\$2,265.82
312	262	18	00	\$2,265.82
312	262	19	00	\$2,265.82
312	262	20	00	\$2,265.82
312	262	21	00	\$2,265.82
312	262	22	00	\$2,265.82
312	262	23	00	\$2,265.82
312	262	24	00	\$2,265.82
312	262	25	00	\$2,265.82
312	262	26	00	\$2,265.82
312	262	27	00	\$2,265.82
312	262	28	00	\$2,265.82
312	262	29	00	\$2,265.82
312	262	30	00	\$2,265.82
312	262	31	00	\$2,265.82
312	262	32	00	\$2,265.82
312	262	33	00	\$2,265.82
312	263	1	00	\$2,265.82
312	263	2	00	\$2,265.82
312	263	3	00	\$2,265.82
312	263	4	00	\$2,253.68
312	263	5	00	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	263	6	00	\$2,253.68
312	263	7	00	\$2,265.82
312	262	34	00	\$2,265.82
312	262	35	00	\$2,265.82
312	262	36	00	\$2,265.82
312	262	37	00	\$2,265.82
312	262	38	00	\$2,265.82
312	262	39	00	\$2,265.82
312	262	40	00	\$2,265.82
312	262	41	00	\$2,265.82
312	262	42	00	\$2,265.82
312	262	43	00	\$2,265.82
312	262	44	00	\$2,265.82
312	262	45	00	\$2,265.82
312	263	8	00	\$2,265.82
312	263	9	00	\$2,265.82
312	263	10	00	\$2,265.82
312	263	11	00	\$2,265.82
312	263	12	00	\$2,265.82
312	263	13	00	\$2,265.82
312	263	14	00	\$2,253.68
312	263	15	00	\$2,253.68
312	263	16	00	\$2,253.68
312	263	17	00	\$2,253.68
312	263	18	00	\$2,253.68
312	263	19	00	\$2,253.68
312	263	20	00	\$2,253.68
312	263	21	00	\$2,253.68
312	263	22	00	\$2,253.68
312	263	23	00	\$2,253.68
312	263	24	00	\$2,253.68
312	263	25	00	\$2,253.68
312	263	26	00	\$2,253.68
312	263	27	00	\$2,253.68
312	263	28	00	\$2,265.82
312	263	29	00	\$2,265.82
312	263	30	00	\$2,265.82
312	263	31	00	\$2,265.82
312	263	32	00	\$2,265.82
312	263	33	00	\$2,265.82
312	263	34	00	\$2,265.82
312	263	35	00	\$2,265.82

	SPECIAL			
воок	PAGE	PARCEL	INT	TAX
312	263	36	00	\$2,265.82
312	263	37	00	\$2,265.82
312	263	38	00	\$2,265.82
312	263	39	00	\$2,265.82
312	263	40	00	\$2,265.82
312	263	41	00	\$2,265.82
312	264	1	00	\$2,265.82
312	264	2	00	\$2,265.82
312	264	3	00	\$2,265.82
312	264	4	00	\$2,265.82
312	264	5	00	\$2,265.82
312	264	6	00	\$2,265.82
312	264	7	00	\$2,265.82
312	264	8	00	\$2,265.82
312	264	9	00	\$2,265.82
312	264	10	00	\$2,265.82
312	264	11	00	\$2,265.82
312	264	12	00	\$2,265.82
312	264	13	00	\$2,265.82
312	264	14	00	\$2,265.82
312	264	15	00	\$2,265.82
312	264	16	00	\$2,265.82
312	264	17	00	\$2,265.82
312	264	18	00	\$2,265.82
312	264	19	00	\$2,265.82
312	264	20	00	\$2,265.82
312	264	21	00	\$2,265.82
312	264	22	00	\$2,265.82
312	264	23	00	\$2,265.82
312	264	24	00	\$2,265.82
312	264	25	00	\$2,265.82
312	264	26	00	\$2,265.82
312	264	27	00	\$2,265.82
312	265	1	00	\$2,265.82
312	265	2	00	\$2,265.82
312	265	3	00	\$2,265.82
312	265	4	00	\$2,265.82
312	265	5	00	\$2,265.82
312	265	6	00	\$2,265.82
312	265	7	00	\$2,265.82
312	265	8	00	\$2,265.82
312	265	9	00	\$2,265.82

	ASSESSOR PA	RCEL NUMBER	L	SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	265	10	00	\$2,265.82
312	265	11	00	\$2,265.82
312	265	12	00	\$2,265.82
312	265	13	00	\$2,265.82
312	265	14	00	\$2,265.82
312	265	15	00	\$2,265.82
312	265	16	00	\$2,265.82
312	265	17	00	\$2,265.82
312	265	18	00	\$2,265.82
312	265	19	00	\$2,265.82
312	265	20	00	\$2,265.82
312	265	21	00	\$2,265.82
312	264	28	00	\$2,265.82
312	264	29	00	\$2,265.82
312	264	30	00	\$2,265.82
312	264	31	00	\$2,265.82
312	264	32	00	\$2,265.82
312	264	33	00	\$2,265.82
312	264	34	00	\$2,265.82
312	264	35	00	\$2,265.82
312	264	36	00	\$2,265.82
312	264	37	00	\$2,265.82
312	264	38	00	\$2,265.82
312	264	39	00	\$2,265.82
312	264	40	00	\$2,253.68
312	264	41	00	\$2,253.68
312	264	42	00	\$2,253.68
312	264	43	00	\$2,265.82
312	260	50	00	\$0.00
312	265	22	00	\$0.00
312	262	46	00	\$0.00
312	262	47	00	\$0.00
312	270	1	00	\$2,265.82
312	270	2	00	\$2,265.82
312	270	3	00	\$2,265.82
312	270	4	00	\$2,265.82
312	270	5	00	\$2,265.82
312	270	6	00	\$2,265.82
312	270	7	00	\$2,265.82
312	270	8	00	\$2,265.82
312	270	9	00	\$2,265.82
312	270	10	00	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	270	11	00	\$2,265.82
312	270	12	00	\$2,265.82
312	270	13	00	\$2,265.82
312	270	14	00	\$2,265.82
312	270	15	00	\$2,265.82
312	270	16	00	\$2,265.82
312	270	17	00	\$2,265.82
312	270	18	00	\$2,265.82
312	270	19	00	\$2,265.82
312	270	20	00	\$2,265.82
312	270	21	00	\$2,265.82
312	270	22	00	\$2,265.82
312	270	23	00	\$2,265.82
312	270	24	00	\$2,265.82
312	270	25	00	\$2,265.82
312	270	26	00	\$2,265.82
312	270	27	00	\$2,265.82
312	270	28	00	\$2,265.82
312	270	29	00	\$2,265.82
312	270	30	00	\$2,265.82
312	270	31	00	\$2,265.82
312	270	32	00	\$2,265.82
312	270	33	00	\$2,265.82
312	270	34	00	\$2,265.82
312	270	35	00	\$2,265.82
312	270	36	00	\$2,265.82
312	270	37	00	\$2,265.82
312	270	38	00	\$2,265.82
312	270	39	00	\$2,265.82
312	270	40	00	\$2,265.82
312	270	41	00	\$2,265.82
312	270	42	00	\$2,265.82
312	270	43	00	\$2,265.82
312	270	44	00	\$2,265.82
312	270	45	00	\$2,265.82
312	270	46	00	\$2,265.82
312	270	47	00	\$0.00
312	270	48	00	\$0.00
312	270	49	00	\$0.00
312	271	1	00	\$2,265.82
312	271	2	00	\$2,265.82
312	271	3	00	\$2,265.82

	ASSESSOR PARCEL NUMBER					
воок	PAGE	PARCEL	INT	TAX		
312	271	4	00	\$2,265.82		
312	271	5	00	\$2,265.82		
312	271	6	00	\$2,265.82		
312	271	7	00	\$2,265.82		
312	271	8	00	\$2,265.82		
312	271	9	00	\$2,265.82		
312	271	10	00	\$2,265.82		
312	271	11	00	\$2,265.82		
312	271	12	00	\$2,265.82		
312	271	13	00	\$2,265.82		
312	271	14	00	\$2,265.82		
312	271	15	00	\$2,265.82		
312	271	16	00	\$2,265.82		
312	271	17	00	\$2,265.82		
312	271	18	00	\$2,265.82		
312	271	19	00	\$2,265.82		
312	271	20	00	\$2,265.82		
312	271	21	00	\$2,265.82		
312	271	22	00	\$2,265.82		
312	271	23	00	\$2,265.82		
312	271	24	00	\$2,265.82		
312	271	25	00	\$2,265.82		
312	271	26	00	\$2,265.82		
312	271	27	00	\$2,265.82		
312	271	28	00	\$2,265.82		
312	271	29	00	\$2,265.82		
312	271	30	00	\$2,265.82		
312	271	31	00	\$2,265.82		
312	271	32	00	\$2,265.82		
312	271	33	00	\$2,265.82		
312	271	34	00	\$2,265.82		
312	271	35	00	\$2,265.82		
312	271	36	00	\$0.00		
312	271	37	00	\$0.00		
312	272	1	00	\$2,265.82		
312	272	2	00	\$2,265.82		
312	272	3	00	\$2,265.82		
312	272	4	00	\$2,265.82		
312	272	5	00	\$2,265.82		
312	272	6	00	\$2,265.82		
312	272	7	00	\$2,265.82		
312	272	8	00	\$2,265.82		

	ASSESSOR PA	RCEL NUMBER		SPECIAL
ВООК	PAGE	PARCEL	INT	TAX
312	272	9	00	\$2,265.82
312	272	10	00	\$2,265.82
312	272	11	00	\$2,265.82
312	272	12	00	\$2,265.82
312	272	13	00	\$2,265.82
312	272	14	00	\$2,265.82
312	272	15	00	\$2,265.82
312	272	16	00	\$2,265.82
312	272	17	00	\$2,265.82
312	272	18	00	\$2,265.82
312	272	19	00	\$2,265.82
312	272	20	00	\$2,265.82
312	272	21	00	\$2,265.82
312	272	22	00	\$2,265.82
312	272	23	00	\$2,265.82
312	272	24	00	\$2,265.82
312	272	25	00	\$2,430.44
312	272	26	00	\$2,430.44
312	272	27	00	\$2,430.44
312	272	28	00	\$2,430.44
312	272	29	00	\$2,430.44
312	272	30	00	\$2,265.82
312	272	31	00	\$2,265.82
312	272	32	00	\$2,265.82
312	272	33	00	\$2,265.82
312	272	34	00	\$2,265.82
312	272	35	00	\$2,265.82
312	272	36	00	\$2,265.82
312	272	37	00	\$2,265.82
312	272	38	00	\$2,265.82
312	272	39	00	\$0.00
312	273	1	00	\$2,265.82
312	273	2	00	\$2,265.82
312	273	3	00	\$2,265.82
312	273	4	00	\$2,430.44
312	273	5	00	\$2,430.44
312	273	6	00	\$2,265.82
312	273	7	00	\$2,265.82
312	273	8	00	\$2,265.82
312	273	9	00	\$2,265.82
312	273	10	00	\$2,265.82
312	273	11	00	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	273	12	00	\$2,265.82
312	273	13	00	\$2,265.82
312	273	14	00	\$2,265.82
312	273	15	00	\$2,265.82
312	273	16	00	\$2,265.82
312	273	17	00	\$2,265.82
312	273	18	00	\$2,265.82
312	273	19	00	\$2,265.82
312	273	20	00	\$2,265.82
312	273	21	00	\$2,265.82
312	273	22	00	\$2,265.82
312	273	23	00	\$2,265.82
312	273	24	00	\$2,265.82
312	273	25	00	\$2,265.82
312	273	26	00	\$2,265.82
312	273	27	00	\$2,265.82
312	273	28	00	\$2,265.82
312	273	29	00	\$2,265.82
312	273	30	00	\$2,265.82
312	273	31	00	\$2,265.82
312	273	32	00	\$2,265.82
312	273	33	00	\$2,265.82
312	273	34	00	\$2,265.82
312	273	35	00	\$2,265.82
312	273	36	00	\$2,265.82
312	273	37	00	\$0.00
312	273	38	00	\$0.00
312	274	1	00	\$2,265.82
312	274	2	00	\$2,265.82
312	274	3	00	\$2,265.82
312	274	4	00	\$2,265.82
312	274	5	00	\$2,265.82
312	274	6	00	\$2,265.82
312	274	7	00	\$2,265.82
312	274	8	00	\$2,265.82
312	274	9	00	\$2,265.82
312	274	10	00	\$2,265.82
312	274	11	00	\$2,265.82
312	274	12	00	\$2,265.82
312	274	13	00	\$2,265.82
312	274	14	00	\$2,265.82
312	274	15	00	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	274	16	00	\$2,265.82
312	274	17	00	\$2,265.82
312	274	18	00	\$2,265.82
312	274	19	00	\$2,265.82
312	274	20	00	\$2,265.82
312	274	21	00	\$2,265.82
312	274	22	00	\$2,265.82
312	274	23	00	\$2,265.82
312	274	24	00	\$2,265.82
312	274	25	00	\$2,265.82
312	274	26	00	\$2,265.82
312	274	27	00	\$2,265.82
312	274	28	00	\$2,265.82
312	274	29	00	\$2,265.82
312	274	30	00	\$2,265.82
312	274	31	00	\$2,265.82
312	274	32	00	\$2,265.82
312	274	33	00	\$2,265.82
312	274	34	00	\$2,265.82
312	274	35	00	\$2,265.82
312	274	36	00	\$2,265.82
312	274	37	00	\$2,265.82
312	274	38	00	\$2,265.82
312	274	39	00	\$2,265.82
312	274	40	00	\$2,265.82
312	274	41	00	\$2,265.82
312	274	42	00	\$0.00
312	275	1	00	\$2,265.82
312	275	2	00	\$2,265.82
312	275	3	00	\$2,265.82
312	275	4	00	\$2,265.82
312	275	5	00	\$2,265.82
312	275	6	00	\$2,265.82
312	275	7	00	\$2,265.82
312	275	8	00	\$2,265.82
312	275	9	00	\$2,265.82
312	275	10	00	\$2,265.82
312	275	11	00	\$2,265.82
312	275	12	00	\$2,265.82
312	275	13	00	\$2,265.82
312	275	14	00	\$2,265.82
312	275	15	00	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	275	16	00	\$2,265.82
312	275	17	00	\$2,265.82
312	275	18	00	\$2,265.82
312	275	19	00	\$2,265.82
312	275	20	00	\$2,265.82
312	275	21	00	\$2,265.82
312	275	22	00	\$2,265.82
312	275	23	00	\$2,265.82
312	275	24	00	\$2,265.82
312	275	25	00	\$2,265.82
312	275	26	00	\$0.00
312	275	27	00	\$0.00
312	275	28	00	\$0.00
312	276	1	00	\$2,265.82
312	276	2	00	\$2,265.82
312	276	3	00	\$2,265.82
312	276	4	00	\$2,265.82
312	276	5	00	\$2,265.82
312	276	6	00	\$2,265.82
312	276	7	00	\$2,265.82
312	276	8	00	\$2,265.82
312	276	9	00	\$2,265.82
312	276	10	00	\$2,265.82
312	276	11	00	\$2,265.82
312	276	12	00	\$2,265.82
312	276	13	00	\$2,265.82
312	276	14	00	\$2,265.82
312	276	15	00	\$2,265.82
312	276	16	00	\$2,265.82
312	276	17	00	\$2,265.82
312	276	18	00	\$2,265.82
312	276	19	00	\$2,265.82
312	276	20	00	\$2,265.82
312	276	21	00	\$2,265.82
312	276	22	00	\$2,265.82
312	276	23	00	\$2,265.82
312	276	24	00	\$2,265.82
312	276	25	00	\$0.00
312	276	26	00	\$0.00
312	276	27	00	\$0.00
312	280	1	00	\$0.00
312	280	2	00	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK PAGE PARCEL I				TAX
312	280	3	00	\$0.00
312	280	4	00	\$0.00
312	280	5	00	\$0.00
312	280	6	00	\$0.00
312	280	7	00	\$0.00
312	280	8	00	\$0.00
312	280	9	00	\$0.00
312	280	10	00	\$0.00
312	280	11	00	\$0.00
312	280	12	00	\$0.00
312	280	13	00	\$0.00
312	280	14	00	\$0.00
312	280	15	00	\$0.00
312	280	16	00	\$0.00
312	280	17	00	\$0.00
312	280	18	00	\$0.00
312	280	19	00	\$0.00
312	280	20	00	\$0.00
312	280	21	00	\$0.00
312	280	22	00	\$0.00
312	280	23	00	\$0.00
312	280	24	00	\$0.00
312	280	25	00	\$0.00
312	280	26	00	\$0.00
312	281	1	00	\$0.00
312	281	2	00	\$0.00
312	281	3	00	\$0.00
312	281	4	00	\$0.00
312	281	5	00	\$0.00
312	281	6	00	\$0.00
312	281	7	00	\$0.00
312	281	8	00	\$0.00
312	282	1	00	\$2,462.36
312	282	2	00	\$0.00
312	282	3	00	\$0.00
312	282	4	00	\$2,462.36
312	282	5	00	\$2,462.36
312	282	6	00	\$2,462.36
312	282	7	00	\$2,462.36
312	282	8	00	\$2,462.36
312	282	9	00	\$2,462.36
312	282	10	00	\$2,462.36

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
312	282	11	00	\$2,462.36
312	282	12	00	\$2,462.36
312	282	13	00	\$2,462.36
312	282	14	00	\$2,462.36
312	282	15	00	\$2,462.36
312	282	16	00	\$2,462.36
312	282	17	00	\$2,462.36
312	282	18	00	\$0.00
312	282	19	00	\$0.00
312	282	20	00	\$0.00
312	282	21	00	\$0.00
312	282	22	00	\$0.00
312	282	23	00	\$0.00
312	282	24	00	\$0.00
312	282	25	00	\$0.00
312	282	26	00	\$0.00
312	282	27	00	\$0.00
312	282	28	00	\$0.00
312	282	29	00	\$0.00
312	282	30	00	\$0.00
312	282	31	00	\$0.00
312	282	32	00	\$0.00
312	282	33	00	\$0.00
312	282	34	00	\$0.00
312	282	35	00	\$0.00
312	282	36	00	\$0.00
312	283	1	00	\$2,462.36
312	283	2	00	\$2,462.36
312	283	3	00	\$2,462.36
312	283	4	00	\$2,462.36
312	283	5	00	\$2,462.36
312	283	6	00	\$2,462.36
312	283	7	00	\$2,462.36
312	283	8	00	\$2,462.36
312	283	9	00	\$2,462.36
312	283	10	00	\$2,462.36
312	283	11	00	\$2,462.36
312	283	12	00	\$2,462.36
312	283	13	00	\$2,462.36
312	283	14	00	\$2,462.36
312	283	15	00	\$2,462.36
312	283	16	00	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	BOOK PAGE PARCEL INT			
312	283	17	00	\$0.00
312	283	18	00	\$0.00
312	283	19	00	\$0.00
312	283	20	00	\$0.00
312	283	21	00	\$0.00
312	283	22	00	\$0.00
312	283	23	00	\$2,462.36
312	283	24	00	\$2,462.36
312	283	25	00	\$0.00
312	283	26	00	\$0.00
312	283	27	00	\$0.00
312	283	28	00	\$0.00
312	283	29	00	\$0.00
312	283	30	00	\$0.00
312	283	31	00	\$0.00
312	283	32	00	\$0.00
312	283	33	00	\$0.00
312	283	34	00	\$0.00
312	283	35	00	\$0.00
312	283	36	00	\$0.00
312	284	1	00	\$0.00
312	284	2	00	\$0.00
312	284	3	00	\$0.00
312	284	4	00	\$0.00
312	285	1	00	\$0.00
678	638	6	0	\$0.00
678	640	1	0	\$0.00
678	640	2	0	\$0.00
678	640	3	0	\$0.00
678	640	4	0	\$0.00
678	640	5	0	\$0.00
678	640	6	0	\$0.00
678	640	7	0	\$0.00
678	640	8	0	\$0.00
678	640	9	0	\$0.00
678	640	10	0	\$0.00
678	640	11	0	\$0.00
678	640	12	0	\$0.00
678	640	13	0	\$0.00
678	640	14	0	\$0.00
678	640	15	0	\$0.00
678	640	16	0	\$0.00

	SPECIAL				
воок	BOOK PAGE PARCEL INT				
678	640	17	0	\$0.00	
678	640	18	0	\$0.00	
678	640	19	0	\$0.00	
678	640	20	0	\$0.00	
678	640	21	0	\$0.00	
678	640	22	0	\$0.00	
678	640	23	0	\$0.00	
678	640	24	00	\$0.00	
678	640	25	00	\$0.00	
678	640	26	00	\$0.00	
678	640	27	00	\$0.00	
678	640	28	00	\$0.00	
678	640	29	00	\$0.00	
678	640	30	00	\$0.00	
678	640	31	00	\$0.00	
678	640	32	00	\$0.00	
678	640	33	00	\$0.00	
678	640	34	00	\$0.00	
678	640	35	00	\$0.00	
678	640	36	00	\$0.00	
678	640	37	00	\$0.00	
678	640	38	00	\$0.00	
678	640	39	00	\$0.00	
678	640	40	00	\$0.00	
678	640	41	00	\$0.00	
678	640	42	00	\$0.00	
678	640	43	00	\$0.00	
678	640	44	00	\$0.00	
678	640	45	00	\$0.00	
678	640	46	00	\$0.00	
678	640	47	00	\$0.00	
678	640	48	00	\$0.00	
678	640	49	00	\$0.00	
678	640	50	00	\$0.00	
678	640	51	00	\$0.00	
678	640	52	00	\$0.00	
678	640	53	00	\$0.00	
678	640	54	00	\$0.00	
678	640	55	00	\$0.00	
678	640	56	00	\$0.00	
678	640	57	00	\$0.00	
678	640	58	00	\$0.00	

	ASSESSOR PARCEL NUMBER				
воок	BOOK PAGE PARCEL INT				
678	640	59	00	\$0.00	
678	640	60	00	\$0.00	
678	640	61	00	\$0.00	
678	640	62	00	\$0.00	
678	640	63	00	\$0.00	
678	640	64	00	\$0.00	
678	640	65	00	\$0.00	
678	640	66	00	\$0.00	
678	640	67	00	\$0.00	
678	640	68	00	\$0.00	
678	641	1	00	\$0.00	
678	641	2	00	\$0.00	
678	641	3	00	\$0.00	
678	641	4	00	\$0.00	
678	641	5	00	\$0.00	
678	641	6	00	\$0.00	
678	641	7	00	\$0.00	
678	641	8	00	\$0.00	
678	641	9	00	\$0.00	
678	641	10	00	\$0.00	
678	641	11	00	\$0.00	
678	641	12	00	\$0.00	
678	641	13	00	\$0.00	
678	641	14	00	\$0.00	
678	641	15	00	\$0.00	
678	641	16	00	\$0.00	
678	641	17	00	\$0.00	
678	641	18	00	\$0.00	
678	641	19	00	\$0.00	
678	641	20	00	\$0.00	
678	641	21	00	\$0.00	
678	641	22	00	\$0.00	
678	641	23	00	\$0.00	
678	641	24	00	\$0.00	
678	641	25	00	\$0.00	
678	641	26	00	\$0.00	
678	641	27	00	\$0.00	
678	641	28	00	\$0.00	
678	641	29	00	\$0.00	
678	641	30	00	\$0.00	
678	641	31	00	\$0.00	
678	641	32	00	\$0.00	

	ASSESSOR PARCEL NUMBER				
воок	BOOK PAGE PARCEL INT				
678	641	33	00	\$0.00	
678	641	34	00	\$0.00	
678	641	35	00	\$0.00	
678	641	36	00	\$0.00	
678	641	37	00	\$0.00	
678	641	38	00	\$0.00	
678	641	39	00	\$0.00	
678	641	40	00	\$0.00	
678	641	41	00	\$0.00	
678	641	42	00	\$0.00	
678	641	43	00	\$0.00	
678	641	44	00	\$0.00	
678	641	45	00	\$0.00	
678	641	46	00	\$0.00	
678	641	47	00	\$0.00	
678	641	48	00	\$0.00	
678	641	49	00	\$0.00	
678	641	50	00	\$0.00	
678	641	51	00	\$0.00	
678	641	52	00	\$0.00	
678	641	53	00	\$0.00	
678	641	54	00	\$0.00	
678	641	55	00	\$0.00	
678	642	1	00	\$0.00	
678	642	2	00	\$0.00	
678	642	3	00	\$0.00	
678	642	4	00	\$0.00	
678	642	5	00	\$0.00	
678	642	6	00	\$0.00	
678	642	7	00	\$0.00	
678	642	8	00	\$0.00	
678	642	9	00	\$0.00	
678	642	10	00	\$0.00	
678	642	11	00	\$0.00	
678	642	12	00	\$0.00	
678	642	13	00	\$0.00	
678	642	14	00	\$0.00	
678	642	15	00	\$0.00	
678	642	16	00	\$0.00	
678	642	17	00	\$0.00	
678	642	18	00	\$0.00	
678	642	19	00	\$0.00	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	642	20	00	\$0.00
678	642	21	00	\$0.00
678	642	22	00	\$0.00
678	642	23	00	\$0.00
678	642	24	00	\$0.00
678	642	25	00	\$0.00
678	642	26	00	\$0.00
678	642	27	00	\$0.00
678	642	28	00	\$0.00
678	642	29	00	\$0.00
678	642	30	00	\$0.00
678	642	31	00	\$0.00
678	642	32	00	\$0.00
678	642	33	00	\$0.00
678	642	34	00	\$0.00
678	642	35	00	\$0.00
678	642	36	00	\$0.00
678	642	37	00	\$0.00
678	642	38	00	\$0.00
678	642	39	00	\$0.00
678	642	40	00	\$0.00
678	642	41	00	\$0.00
678	642	42	00	\$0.00
678	642	43	00	\$0.00
678	642	44	00	\$0.00
678	642	45	00	\$0.00
678	642	46	00	\$0.00
678	642	47	00	\$0.00
678	643	1	00	\$0.00
678	643	2	00	\$0.00
678	643	3	00	\$0.00
678	643	4	00	\$0.00
678	643	5	00	\$0.00
678	643	6	00	\$0.00
678	643	7	00	\$0.00
678	643	8	00	\$0.00
678	643	9	00	\$0.00
678	643	10	00	\$0.00
678	643	11	00	\$0.00
678	643	12	00	\$0.00
678	643	13	00	\$0.00
678	643	14	00	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	BOOK PAGE PARCEL INT			
678	643	15	00	\$0.00
678	643	16	00	\$0.00
678	643	17	00	\$0.00
678	643	18	00	\$0.00
678	643	19	00	\$0.00
678	643	20	00	\$0.00
678	643	21	00	\$0.00
678	643	22	00	\$0.00
678	643	23	00	\$0.00
678	643	24	00	\$0.00
678	643	25	00	\$0.00
678	643	26	00	\$0.00
678	643	27	00	\$0.00
678	643	28	00	\$0.00
678	643	29	00	\$0.00
678	643	30	00	\$0.00
678	643	31	00	\$0.00
678	643	32	00	\$0.00
678	643	33	00	\$0.00
678	643	34	00	\$0.00
678	643	35	00	\$0.00
678	643	36	00	\$0.00
678	643	37	00	\$0.00
678	643	38	00	\$0.00
678	643	39	00	\$0.00
678	643	40	00	\$0.00
678	643	41	00	\$0.00
678	643	42	00	\$0.00
678	643	43	00	\$0.00
678	643	44	00	\$0.00
678	643	45	00	\$0.00
678	643	46	00	\$0.00
678	643	47	00	\$0.00
678	643	48	00	\$0.00
678	643	49	00	\$0.00
678	643	50	00	\$0.00
678	644	1	00	\$0.00
678	644	2	00	\$0.00
678	644	3	00	\$0.00
678	644	4	00	\$0.00
678	644	5	00	\$0.00
678	644	6	00	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL	
воок	BOOK PAGE PARCEL INT				
678	644	7	00	\$0.00	
678	644	8	00	\$0.00	
678	644	9	00	\$0.00	
678	644	10	00	\$0.00	
678	644	11	00	\$0.00	
678	644	12	00	\$0.00	
678	644	13	00	\$0.00	
678	644	14	00	\$0.00	
678	644	15	00	\$0.00	
678	644	16	00	\$0.00	
678	644	17	00	\$0.00	
678	644	18	00	\$0.00	
678	644	19	00	\$0.00	
678	644	20	00	\$0.00	
678	644	21	00	\$0.00	
678	644	22	00	\$0.00	
678	644	23	00	\$0.00	
678	644	24	00	\$0.00	
678	644	25	00	\$0.00	
678	644	26	00	\$0.00	
678	644	27	00	\$0.00	
678	644	28	00	\$0.00	
678	644	29	00	\$0.00	
678	644	30	00	\$0.00	
678	644	31	00	\$0.00	
678	644	32	00	\$0.00	
678	644	33	00	\$0.00	
678	644	34	00	\$0.00	
678	644	35	00	\$0.00	
678	644	36	00	\$0.00	
678	644	37	00	\$0.00	
678	644	38	00	\$0.00	
678	644	39	00	\$0.00	
678	644	40	00	\$0.00	
678	644	41	00	\$0.00	
678	644	42	00	\$0.00	
678	630	01	00	\$0.00	
678	630	02	00	\$0.00	
678	630	03	00	\$0.00	
678	630	04	00	\$0.00	
678	630	05	00	\$0.00	
678	630	06	00	\$0.00	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	630	07	00	\$0.00
678	630	08	00	\$0.00
678	630	09	00	\$0.00
678	630	10	00	\$0.00
678	630	11	00	\$0.00
678	630	12	00	\$0.00
678	630	13	00	\$0.00
678	630	14	00	\$0.00
678	630	89	00	\$0.00
678	630	90	00	\$0.00
678	630	17	00	\$0.00
678	630	18	00	\$0.00
678	630	19	00	\$0.00
678	630	20	00	\$0.00
678	630	21	00	\$2,462.36
678	630	22	00	\$2,462.36
678	630	23	00	\$2,462.36
678	630	24	00	\$2,462.36
678	630	25	00	\$2,462.36
678	630	26	00	\$2,462.36
678	630	27	00	\$2,462.36
678	630	28	00	\$2,462.36
678	630	29	00	\$2,462.36
678	630	30	00	\$2,462.36
678	630	31	00	\$2,462.36
678	630	32	00	\$2,462.36
678	630	33	00	\$2,462.36
678	630	34	00	\$2,462.36
678	630	35	00	\$2,462.36
678	630	36	00	\$2,462.36
678	630	37	00	\$2,462.36
678	630	38	00	\$2,462.36
678	630	87	00	\$2,462.36
678	630	88	00	\$2,462.36
678	630	41	00	\$0.00
678	630	42	00	\$0.00
678	630	43	00	\$0.00
678	630	44	00	\$0.00
678	630	45	00	\$0.00
678	630	46	00	\$0.00
678	630	47	00	\$0.00
678	630	48	00	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL		
воок	BOOK PAGE PARCEL INT					
678	630	49	00	\$0.00		
678	630	50	00	\$0.00		
678	630	51	00	\$2,462.36		
678	630	52	00	\$2,462.36		
678	630	53	00	\$2,462.36		
678	630	54	00	\$2,462.36		
678	630	55	00	\$2,462.36		
678	630	56	00	\$2,462.36		
678	630	57	00	\$2,462.36		
678	630	58	00	\$2,462.36		
678	630	59	00	\$2,462.36		
678	630	60	00	\$2,462.36		
678	630	61	00	\$0.00		
678	630	62	00	\$2,462.36		
678	630	63	00	\$2,462.36		
678	630	64	00	\$2,462.36		
678	630	65	00	\$2,462.36		
678	630	66	00	\$0.00		
678	630	67	00	\$2,462.36		
678	630	68	00	\$2,462.36		
678	630	69	00	\$2,462.36		
678	630	70	00	\$2,462.36		
678	630	71	00	\$2,462.36		
678	630	72	00	\$2,462.36		
678	630	73	00	\$2,462.36		
678	630	74	00	\$2,462.36		
678	630	75	00	\$2,462.36		
678	630	76	00	\$2,462.36		
678	630	77	00	\$2,462.36		
678	630	78	00	\$2,462.36		
678	630	79	00	\$2,462.36		
678	630	80	00	\$2,462.36		
678	630	81	00	\$2,462.36		
678	630	82	00	\$2,462.36		
678	630	83	00	\$2,462.36		
678	630	84	00	\$2,462.36		
678	630	85	00	\$2,462.36		
678	630	86	00	\$2,462.36		
678	631	01	00	\$0.00		
678	631	02	00	\$0.00		
678	631	03	00	\$0.00		
678	631	04	00	\$0.00		

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	631	05	00	\$0.00
678	631	06	00	\$0.00
678	631	07	00	\$0.00
678	631	08	00	\$0.00
678	631	09	00	\$0.00
678	631	10	00	\$0.00
678	631	11	00	\$0.00
678	631	12	00	\$0.00
678	631	13	00	\$0.00
678	631	14	00	\$0.00
678	631	15	00	\$0.00
678	631	16	00	\$0.00
678	631	17	00	\$0.00
678	631	18	00	\$0.00
678	631	19	00	\$0.00
678	631	20	00	\$0.00
678	631	21	00	\$0.00
678	631	22	00	\$2,430.44
678	631	23	00	\$2,430.44
678	631	24	00	\$2,430.44
678	631	25	00	\$2,430.44
678	631	26	00	\$2,430.44
678	631	27	00	\$2,430.44
678	631	28	00	\$2,430.44
678	631	29	00	\$2,462.36
678	631	30	00	\$2,462.36
678	631	31	00	\$2,462.36
678	631	32	00	\$2,430.44
678	631	33	00	\$2,430.44
678	631	34	00	\$2,430.44
678	631	35	00	\$2,430.44
678	631	36	00	\$2,430.44
678	631	37	00	\$0.00
678	631	38	00	\$0.00
678	631	39	00	\$0.00
678	631	40	00	\$0.00
678	631	41	00	\$0.00
678	631	42	00	\$0.00
678	631	43	00	\$0.00
678	631	44	00	\$0.00
678	631	45	00	\$0.00
678	631	46	00	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL	
воок	BOOK PAGE PARCEL INT				
678	631	47	00	\$0.00	
678	631	48	00	\$0.00	
678	631	49	00	\$0.00	
678	631	50	00	\$0.00	
678	632	01	00	\$2,430.44	
678	632	02	00	\$2,430.44	
678	632	03	00	\$2,430.44	
678	632	04	00	\$2,430.44	
678	632	05	00	\$2,430.44	
678	632	06	00	\$2,430.44	
678	632	07	00	\$2,430.44	
678	632	08	00	\$2,430.44	
678	632	09	00	\$0.00	
678	632	10	00	\$0.00	
678	632	11	00	\$0.00	
678	632	12	00	\$0.00	
678	632	13	00	\$0.00	
678	632	14	00	\$0.00	
678	632	15	00	\$0.00	
678	632	16	00	\$0.00	
678	632	17	00	\$0.00	
678	632	18	00	\$2,430.44	
678	632	19	00	\$2,430.44	
678	632	20	00	\$2,430.44	
678	632	21	00	\$0.00	
678	632	22	00	\$0.00	
678	632	23	00	\$0.00	
678	632	24	00	\$0.00	
678	632	25	00	\$2,462.36	
678	632	26	00	\$2,462.36	
678	632	27	00	\$2,462.36	
678	632	28	00	\$2,462.36	
678	632	29	00	\$0.00	
678	632	30	00	\$0.00	
678	632	31	00	\$0.00	
678	632	32	00	\$0.00	
678	632	33	00	\$0.00	
678	632	34	00	\$0.00	
678	632	35	00	\$0.00	
678	632	36	00	\$0.00	
678	632	37	00	\$2,462.36	
678	632	38	00	\$2,462.36	

	ASSESSOR PARCEL NUMBER				
воок	BOOK PAGE PARCEL INT				
678	632	39	00	\$2,462.36	
678	632	40	00	\$2,462.36	
678	632	41	00	\$2,462.36	
678	632	42	00	\$2,462.36	
678	632	43	00	\$2,462.36	
678	632	44	00	\$2,462.36	
678	632	45	00	\$2,462.36	
678	632	46	00	\$2,462.36	
678	632	47	00	\$2,462.36	
678	632	48	00	\$2,462.36	
678	633	01	00	\$2,430.44	
678	633	02	00	\$2,430.44	
678	633	03	00	\$2,430.44	
678	633	04	00	\$2,462.36	
678	633	05	00	\$2,462.36	
678	633	06	00	\$2,430.44	
678	633	07	00	\$2,430.44	
678	633	08	00	\$2,430.44	
678	633	09	00	\$2,430.44	
678	633	10	00	\$2,430.44	
678	633	11	00	\$2,430.44	
678	633	12	00	\$2,430.44	
678	633	13	00	\$2,430.44	
678	633	14	00	\$2,430.44	
678	633	15	00	\$2,430.44	
678	633	16	00	\$2,430.44	
678	633	17	00	\$2,430.44	
678	633	18	00	\$2,430.44	
678	633	19	00	\$2,430.44	
678	633	20	00	\$2,430.44	
678	633	21	00	\$2,430.44	
678	633	22	00	\$2,430.44	
678	633	23	00	\$2,430.44	
678	633	24	00	\$2,430.44	
678	633	25	00	\$2,430.44	
678	633	26	00	\$2,430.44	
678	633	27	00	\$2,430.44	
678	633	28	00	\$2,430.44	
678	633	29	00	\$2,430.44	
678	633	30	00	\$2,430.44	
678	633	31	00	\$2,462.36	
678	633	32	00	\$2,462.36	

	ASSESSOR PA	RCEL NUMBER		SPECIAL	
воок	BOOK PAGE PARCEL INT				
678	633	33	00	\$2,462.36	
678	633	34	00	\$2,462.36	
678	633	35	00	\$2,462.36	
678	633	36	00	\$2,462.36	
678	633	37	00	\$2,462.36	
678	633	38	00	\$2,462.36	
678	633	39	00	\$2,462.36	
678	633	40	00	\$2,462.36	
678	633	41	00	\$2,462.36	
678	633	42	00	\$2,462.36	
678	633	43	00	\$2,462.36	
678	633	44	00	\$2,462.36	
678	633	45	00	\$2,462.36	
678	633	46	00	\$2,462.36	
678	633	47	00	\$2,462.36	
678	633	48	00	\$2,462.36	
678	633	49	00	\$2,462.36	
678	633	50	00	\$2,462.36	
678	633	51	00	\$0.00	
678	633	52	00	\$0.00	
678	633	53	00	\$0.00	
678	633	54	00	\$0.00	
678	633	55	00	\$0.00	
678	633	56	00	\$0.00	
678	633	57	00	\$0.00	
678	633	58	00	\$0.00	
678	633	59	00	\$0.00	
678	633	60	00	\$0.00	
678	634	01	00	\$2,462.36	
678	634	02	00	\$2,462.36	
678	634	03	00	\$2,462.36	
678	634	04	00	\$2,462.36	
678	634	05	00	\$2,462.36	
678	634	06	00	\$2,462.36	
678	634	07	00	\$2,462.36	
678	634	08	00	\$0.00	
678	634	09	00	\$0.00	
678	634	10	00	\$0.00	
678	634	11	00	\$0.00	
678	634	12	00	\$0.00	
678	634	13	00	\$2,462.36	
678	634	14	00	\$2,462.36	

	ASSESSOR PARCEL NUMBER				
воок	TAX				
678	634	15	00	\$2,462.36	
678	634	16	00	\$2,462.36	
678	634	17	00	\$2,462.36	
678	634	18	00	\$2,462.36	
678	634	19	00	\$2,462.36	
678	634	20	00	\$2,462.36	
678	634	21	00	\$2,462.36	
678	634	22	00	\$2,462.36	
678	634	23	00	\$2,462.36	
678	634	24	00	\$2,462.36	
678	634	25	00	\$2,462.36	
678	634	26	00	\$2,462.36	
678	634	27	00	\$2,462.36	
678	634	28	00	\$2,462.36	
678	634	29	00	\$2,462.36	
678	634	30	00	\$2,462.36	
678	634	31	00	\$2,462.36	
678	634	32	00	\$2,462.36	
678	634	33	00	\$2,462.36	
678	634	34	00	\$2,462.36	
678	634	35	00	\$2,462.36	
678	634	36	00	\$2,462.36	
678	634	37	00	\$2,462.36	
678	634	38	00	\$2,462.36	
678	634	39	00	\$2,462.36	
678	634	40	00	\$2,462.36	
678	634	41	00	\$2,462.36	
678	634	42	00	\$2,462.36	
678	634	43	00	\$2,462.36	
678	634	44	00	\$2,462.36	
678	634	45	00	\$2,462.36	
678	634	46	00	\$2,462.36	
678	634	47	00	\$2,462.36	
678	634	48	00	\$0.00	
678	634	49	00	\$0.00	
678	634	50	00	\$0.00	
678	634	51	00	\$0.00	
678	635	01	00	\$0.00	
678	635	02	00	\$0.00	
678	635	03	00	\$0.00	
678	635	04	00	\$0.00	
678	635	05	00	\$0.00	

	ASSESSOR PA	RCEL NUMBER		SPECIAL	
ВООК	BOOK PAGE PARCEL INT				
678	635	06	00	\$0.00	
678	635	07	00	\$0.00	
678	635	08	00	\$2,462.36	
678	635	09	00	\$2,462.36	
678	635	10	00	\$2,462.36	
678	635	11	00	\$2,462.36	
678	635	12	00	\$2,462.36	
678	635	13	00	\$2,462.36	
678	635	14	00	\$2,462.36	
678	635	15	00	\$2,462.36	
678	635	16	00	\$2,462.36	
678	635	17	00	\$2,462.36	
678	635	18	00	\$2,462.36	
678	635	19	00	\$2,462.36	
678	635	20	00	\$2,462.36	
678	635	21	00	\$2,462.36	
678	635	22	00	\$2,462.36	
678	635	23	00	\$2,462.36	
678	635	24	00	\$2,462.36	
678	635	25	00	\$2,462.36	
678	635	26	00	\$2,462.36	
678	635	27	00	\$2,462.36	
678	635	28	00	\$2,462.36	
678	635	29	00	\$2,462.36	
678	635	30	00	\$2,462.36	
678	635	31	00	\$2,462.36	
678	635	32	00	\$2,462.36	
678	635	33	00	\$2,462.36	
678	635	34	00	\$2,462.36	
678	635	35	00	\$2,462.36	
678	635	36	00	\$2,462.36	
678	635	37	00	\$2,462.36	
678	635	38	00	\$2,430.44	
678	635	39	00	\$2,430.44	
678	635	40	00	\$2,430.44	
678	635	41	00	\$2,430.44	
678	635	42	00	\$2,430.44	
678	635	43	00	\$2,462.36	
678	635	44	00	\$2,430.44	
678	635	45	00	\$2,430.44	
678	635	46	00	\$2,430.44	
678	635	47	00	\$2,462.36	

	ASSESSOR PA	RCEL NUMBER		SPECIAL	
воок	BOOK PAGE PARCEL INT				
678	635	48	00	\$2,430.44	
678	635	49	00	\$2,430.44	
678	635	50	00	\$2,430.44	
678	635	51	00	\$2,430.44	
678	635	52	00	\$2,430.44	
678	635	53	00	\$2,430.44	
678	635	54	00	\$2,462.36	
678	635	55	00	\$2,462.36	
678	635	56	00	\$2,462.36	
678	635	57	00	\$2,462.36	
678	635	58	00	\$2,462.36	
678	635	59	00	\$2,462.36	
678	636	01	00	\$2,462.36	
678	636	02	00	\$2,462.36	
678	636	03	00	\$2,462.36	
678	636	04	00	\$2,462.36	
678	636	05	00	\$2,462.36	
678	636	06	00	\$2,462.36	
678	636	07	00	\$2,462.36	
678	636	08	00	\$2,462.36	
678	636	09	00	\$2,462.36	
678	636	10	00	\$2,430.44	
678	636	11	00	\$2,462.36	
678	636	12	00	\$2,462.36	
678	636	13	00	\$2,462.36	
678	636	14	00	\$2,462.36	
678	636	15	00	\$2,462.36	
678	636	16	00	\$2,462.36	
678	636	17	00	\$2,462.36	
678	636	18	00	\$2,462.36	
678	636	19	00	\$2,462.36	
678	636	20	00	\$2,462.36	
678	636	21	00	\$2,462.36	
678	636	22	00	\$2,462.36	
678	636	23	00	\$2,462.36	
678	636	24	00	\$2,462.36	
678	636	25	00	\$2,462.36	
678	636	26	00	\$2,462.36	
678	636	27	00	\$2,462.36	
678	636	28	00	\$2,462.36	
678	636	29	00	\$2,462.36	
678	636	30	00	\$2,462.36	

	SPECIAL			
воок	PAGE	PARCEL	INT	TAX
678	636	31	00	\$2,462.36
678	636	32	00	\$2,462.36
678	636	33	00	\$2,462.36
678	636	34	00	\$2,462.36
678	636	35	00	\$2,462.36
678	636	36	00	\$2,462.36
678	636	37	00	\$2,462.36
678	636	38	00	\$2,462.36
678	636	39	00	\$2,462.36
678	636	40	00	\$2,462.36
678	636	41	00	\$2,462.36
678	636	42	00	\$2,462.36
678	636	43	00	\$2,462.36
678	636	44	00	\$2,462.36
678	636	45	00	\$0.00
678	637	04	00	\$0.00
678	638	04	00	\$0.00
678	638	05	00	\$0.00
678	637	5	1	\$2,462.36
678	637	5	2	\$2,462.36
678	637	5	3	\$2,462.36
678	637	5	4	\$2,462.36
678	637	5	5	\$2,462.36
678	637	5	6	\$2,462.36
678	637	5	7	\$2,462.36
678	637	5	8	\$2,462.36
678	637	5	9	\$2,462.36
678	637	5	10	\$2,462.36
678	637	5	11	\$2,462.36
678	637	5	12	\$2,462.36
678	637	5	13	\$2,462.36
678	637	5	14	\$2,462.36
678	637	5	15	\$2,462.36
678	637	5	16	\$2,462.36
678	637	5	17	\$2,462.36
678	637	5	18	\$2,462.36
678	637	5	19	\$2,462.36
678	637	5	20	\$2,462.36
678	637	5	21	\$2,462.36
678	637	5	22	\$2,462.36
678	637	5	23	\$2,462.36
678	637	5	24	\$2,462.36

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	637	5	25	\$2,462.36
678	637	5	26	\$2,462.36
678	637	5	27	\$2,462.36
678	637	5	28	\$2,462.36
678	637	5	29	\$2,462.36
678	637	5	30	\$2,462.36
678	637	5	31	\$2,462.36
678	637	5	32	\$2,462.36
678	637	5	33	\$2,462.36
678	637	5	34	\$2,462.36
678	637	5	35	\$2,462.36
678	637	5	36	\$2,462.36
678	637	5	37	\$2,462.36
678	637	5	38	\$2,462.36
678	637	5	39	\$2,462.36
678	637	5	40	\$2,462.36
678	637	5	41	\$2,462.36
678	637	5	42	\$2,462.36
678	637	5	43	\$2,462.36
678	637	5	44	\$2,462.36
678	637	5	45	\$2,462.36
678	637	5	46	\$2,462.36
678	637	5	47	\$2,462.36
678	637	5	48	\$2,462.36
678	637	5	49	\$2,462.36
678	637	5	50	\$2,462.36
678	637	5	51	\$2,462.36
678	637	5	52	\$2,462.36
678	637	5	53	\$2,462.36
678	637	5	54	\$2,462.36
678	637	5	55	\$2,462.36
678	637	5	56	\$2,462.36
678	637	5	57	\$2,462.36
678	637	5	58	\$2,462.36
678	637	5	59	\$2,462.36
678	637	5	60	\$2,462.36
678	637	5	61	\$2,462.36
678	637	5	62	\$2,462.36
678	637	5	63	\$2,462.36
678	637	5	64	\$2,462.36
678	637	5	65	\$2,462.36
678	637	5	66	\$2,462.36

	ASSESSOR PARCEL NUMBER					
воок	PAGE	PARCEL	INT	TAX		
678	637	5	67	\$2,462.36		
678	637	5	68	\$2,462.36		
678	637	5	69	\$2,462.36		
678	637	5	70	\$2,462.36		
678	637	5	71	\$0.00		
678	637	5	72	\$0.00		
678	637	5	73	\$0.00		
678	637	6	1	\$0.00		
678	637	6	2	\$2,462.36		
678	637	6	3	\$2,462.36		
678	637	6	4	\$2,462.36		
678	637	6	5	\$2,462.36		
678	637	6	6	\$2,462.36		
678	637	6	7	\$2,462.36		
678	637	6	8	\$2,462.36		
678	637	6	9	\$2,462.36		
678	637	6	10	\$2,462.36		
678	637	6	11	\$2,462.36		
678	637	6	12	\$2,462.36		
678	637	6	13	\$2,462.36		
678	637	6	14	\$2,462.36		
678	637	6	15	\$2,462.36		
678	637	6	16	\$2,462.36		
678	637	6	17	\$2,462.36		
678	637	6	18	\$2,462.36		
678	637	6	19	\$2,462.36		
678	637	6	20	\$2,462.36		
678	637	6	21	\$2,462.36		
678	637	6	22	\$2,462.36		
678	637	6	23	\$2,462.36		
678	637	6	24	\$2,462.36		
678	637	6	25	\$2,462.36		
678	637	6	26	\$2,462.36		
678	637	6	27	\$2,462.36		
678	637	6	28	\$2,462.36		
678	637	6	29	\$2,462.36		
678	637	6	30	\$2,462.36		
678	637	6	31	\$2,462.36		
678	637	6	32	\$2,462.36		
678	637	6	33	\$2,462.36		
678	637	6	34	\$2,462.36		
678	637	6	35	\$2,462.36		

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	637	6	36	\$2,462.36
678	637	6	37	\$2,462.36
678	637	6	38	\$2,462.36
678	637	6	39	\$2,462.36
678	637	6	40	\$2,462.36
678	637	6	41	\$2,462.36
678	637	6	42	\$2,462.36
678	637	6	43	\$2,462.36
678	637	6	44	\$2,462.36
678	637	6	45	\$2,462.36
678	637	6	46	\$2,462.36
678	637	6	47	\$2,462.36
678	637	6	48	\$2,462.36
678	637	6	49	\$2,462.36
678	637	6	50	\$2,462.36
678	637	6	51	\$0.00
678	637	6	52	\$0.00
678	637	6	53	\$0.00
678	637	6	54	\$0.00
678	637	7	1	\$1,089.12
678	637	7	2	\$1,089.12
678	637	7	3	\$1,089.12
678	637	7	4	\$1,089.12
678	637	7	5	\$1,089.12
678	637	7	6	\$1,089.12
678	637	7	7	\$1,089.12
678	637	7	8	\$1,089.12
678	637	7	9	\$1,089.12
678	637	7	10	\$1,089.12
678	637	7	11	\$1,089.12
678	637	7	12	\$1,089.12
678	637	7	13	\$1,089.12
678	637	7	14	\$1,089.12
678	637	7	15	\$1,089.12
678	637	7	16	\$1,089.12
678	637	7	17	\$1,089.12
678	637	7	18	\$1,089.12
678	637	7	19	\$1,089.12
678	637	7	20	\$1,089.12
678	637	7	21	\$1,089.12
678	637	7	22	\$1,089.12
678	637	7	23	\$1,089.12

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	637	7	24	\$1,089.12	
678	637	7	25	\$1,089.12	
678	637	7	26	\$1,089.12	
678	637	7	27	\$1,089.12	
678	637	7	28	\$1,089.12	
678	637	7	29	\$1,089.12	
678	637	7	30	\$1,089.12	
678	637	7	31	\$1,089.12	
678	637	7	32	\$1,089.12	
678	637	7	33	\$1,089.12	
678	637	7	34	\$1,089.12	
678	637	7	35	\$1,089.12	
678	637	7	36	\$1,089.12	
678	637	7	37	\$1,089.12	
678	637	7	38	\$1,089.12	
678	637	7	39	\$1,089.12	
678	637	7	40	\$1,089.12	
678	637	7	41	\$1,089.12	
678	637	7	42	\$1,089.12	
678	637	7	43	\$1,089.12	
678	637	7	44	\$1,089.12	
678	637	7	45	\$1,089.12	
678	637	7	46	\$1,089.12	
678	637	7	47	\$1,089.12	
678	637	7	48	\$1,089.12	
678	637	7	49	\$1,089.12	
678	637	7	50	\$1,089.12	
678	637	7	51	\$1,089.12	
678	637	7	52	\$1,089.12	
678	637	7	53	\$1,089.12	
678	637	7	54	\$1,089.12	
678	637	7	55	\$1,089.12	
678	637	7	56	\$1,089.12	
678	637	7	57	\$1,089.12	
678	637	7	58	\$1,089.12	
678	637	7	59	\$1,089.12	
678	637	7	60	\$1,089.12	
678	637	7	61	\$1,089.12	
678	637	7	62	\$1,089.12	
678	637	7	63	\$1,089.12	
678	637	7	64	\$1,089.12	
678	637	7	65	\$1,089.12	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	637	7	66	\$1,089.12
678	637	7	67	\$1,089.12
678	637	7	68	\$1,089.12
678	637	7	69	\$1,089.12
678	637	7	70	\$1,089.12
678	637	7	71	\$1,089.12
678	637	7	72	\$1,089.12
678	637	7	73	\$1,089.12
678	637	7	74	\$1,089.12
678	637	7	75	\$1,089.12
678	637	7	76	\$1,089.12
678	637	7	77	\$1,089.12
678	637	7	78	\$1,089.12
678	637	7	79	\$1,089.12
678	637	7	80	\$1,089.12
678	637	7	81	\$1,089.12
678	637	7	82	\$1,089.12
678	637	7	83	\$1,089.12
678	637	7	84	\$1,089.12
678	637	7	85	\$1,089.12
678	637	7	86	\$1,089.12
678	637	7	87	\$1,089.12
678	637	7	88	\$1,089.12
678	637	7	89	\$1,089.12
678	637	7	90	\$1,089.12
678	637	8	1	\$1,089.12
678	637	8	2	\$1,089.12
678	637	8	3	\$1,089.12
678	637	8	4	\$1,089.12
678	637	8	5	\$1,089.12
678	637	8	6	\$1,089.12
678	637	8	7	\$1,089.12
678	637	8	8	\$1,089.12
678	637	8	9	\$1,089.12
678	637	8	10	\$1,089.12
678	637	8	11	\$1,089.12
678	637	8	12	\$1,089.12
678	637	8	13	\$1,089.12
678	637	8	14	\$1,089.12
678	637	8	15	\$1,089.12
678	637	8	16	\$1,089.12
678	637	8	17	\$1,089.12

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	637	8	18	\$1,089.12
678	637	8	19	\$1,089.12
678	637	8	20	\$1,089.12
678	637	8	21	\$1,089.12
678	637	8	22	\$1,089.12
678	637	8	23	\$1,089.12
678	637	8	24	\$1,089.12
678	637	8	25	\$1,089.12
678	637	8	26	\$1,089.12
678	637	8	27	\$1,089.12
678	637	8	28	\$1,089.12
678	637	8	29	\$1,089.12
678	637	8	30	\$1,089.12
678	637	8	31	\$1,089.12
678	637	8	32	\$1,089.12
678	637	8	33	\$1,089.12
678	637	8	34	\$1,089.12
678	637	8	35	\$1,089.12
678	637	8	36	\$1,089.12
678	637	8	37	\$1,089.12
678	637	8	38	\$1,089.12
678	637	8	39	\$1,089.12
678	637	8	40	\$1,089.12
678	637	8	41	\$1,089.12
678	637	8	42	\$1,089.12
678	637	8	43	\$1,089.12
678	637	8	44	\$1,089.12
678	637	8	45	\$1,089.12
678	637	8	46	\$1,089.12
678	637	8	47	\$1,089.12
678	637	8	48	\$1,089.12
678	637	8	49	\$1,089.12
678	637	8	50	\$1,089.12
678	637	8	51	\$1,089.12
678	637	8	52	\$1,089.12
678	637	8	53	\$1,089.12
678	637	8	54	\$1,089.12
678	637	8	55	\$1,089.12
678	637	8	56	\$1,089.12
678	637	8	57	\$1,089.12
678	637	8	58	\$1,089.12
678	637	8	59	\$1,089.12

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	637	8	60	\$1,089.12
678	637	8	61	\$1,089.12
678	637	8	62	\$1,089.12
678	637	8	63	\$1,089.12
678	637	8	64	\$1,089.12
678	637	8	65	\$1,089.12
678	637	8	66	\$1,089.12
678	637	8	67	\$1,089.12
678	637	8	68	\$1,089.12
678	637	9	1	\$1,089.12
678	637	9	2	\$1,089.12
678	637	9	3	\$1,089.12
678	637	9	4	\$1,089.12
678	637	9	5	\$1,089.12
678	637	9	6	\$1,089.12
678	637	9	7	\$1,089.12
678	637	9	8	\$1,089.12
678	637	9	9	\$1,089.12
678	637	9	10	\$1,089.12
678	637	9	11	\$1,089.12
678	637	9	12	\$1,089.12
678	637	9	13	\$1,089.12
678	637	9	14	\$1,089.12
678	637	9	15	\$1,089.12
678	637	9	16	\$1,089.12
678	637	9	17	\$1,089.12
678	637	9	18	\$1,089.12
678	637	9	19	\$1,089.12
678	637	9	20	\$1,089.12
678	637	9	21	\$1,089.12
678	637	9	22	\$1,089.12
678	637	9	23	\$1,089.12
678	637	9	24	\$1,089.12
678	637	9	25	\$1,089.12
678	637	9	26	\$1,089.12
678	637	9	27	\$1,089.12
678	637	9	28	\$1,089.12
678	637	9	29	\$1,089.12
678	637	9	30	\$1,089.12
678	637	9	31	\$1,089.12
678	637	9	32	\$1,089.12
678	637	9	33	\$1,089.12

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	637	9	34	\$1,089.12
678	637	9	35	\$1,089.12
678	637	9	36	\$1,089.12
678	637	9	37	\$1,089.12
678	637	9	38	\$1,089.12
678	637	9	39	\$1,089.12
678	638	9	40	\$1,089.12
678	638	9	41	\$1,089.12
678	637	03	0	\$0.00
678	637	03	00	\$0.00
678	638	03	00	\$0.00
678	638	04	00	\$0.00
678	638	05	00	\$0.00
678	502	1	01	\$2,253.68
678	502	1	02	\$2,253.68
678	502	1	03	\$2,253.68
678	502	1	04	\$2,253.68
678	502	1	05	\$2,253.68
678	502	1	06	\$2,253.68
678	502	1	07	\$2,253.68
678	502	1	08	\$2,253.68
678	502	1	09	\$2,253.68
678	502	1	10	\$2,253.68
678	502	1	11	\$2,253.68
678	502	1	12	\$2,253.68
678	502	2	01	\$2,253.68
678	502	2	02	\$2,253.68
678	502	2	03	\$2,253.68
678	502	2	04	\$2,253.68
678	502	2	05	\$2,253.68
678	502	2	06	\$2,253.68
678	502	2	07	\$2,253.68
678	502	2	08	\$2,253.68
678	502	2	09	\$2,253.68
678	502	2	10	\$2,253.68
678	502	2	11	\$2,253.68
678	502	2	12	\$2,253.68
678	502	2	13	\$2,253.68
678	502	2	14	\$2,253.68
678	502	2	15	\$2,253.68
678	502	3	01	\$2,265.82
678	502	3	02	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	502	3	03	\$2,265.82
678	502	3	04	\$2,265.82
678	502	3	05	\$2,265.82
678	502	3	06	\$2,265.82
678	502	3	07	\$2,265.82
678	502	3	08	\$2,265.82
678	502	3	09	\$2,265.82
678	502	3	10	\$2,265.82
678	502	3	11	\$2,253.68
678	502	3	12	\$2,253.68
678	502	3	13	\$2,253.68
678	502	3	14	\$2,253.68
678	502	4	01	\$2,253.68
678	502	4	02	\$2,253.68
678	502	4	03	\$2,253.68
678	502	4	04	\$2,253.68
678	502	4	05	\$2,253.68
678	502	4	06	\$2,253.68
678	502	4	07	\$2,253.68
678	502	4	08	\$2,253.68
678	502	4	09	\$2,253.68
678	502	4	10	\$2,253.68
678	502	4	11	\$2,253.68
678	502	4	12	\$2,253.68
678	502	4	13	\$2,253.68
678	502	4	14	\$2,253.68
678	502	4	15	\$2,253.68
678	502	5	01	\$2,253.68
678	502	5	02	\$2,253.68
678	502	5	03	\$2,253.68
678	502	5	04	\$2,253.68
678	502	5	05	\$2,253.68
678	502	5	06	\$2,253.68
678	502	5	07	\$2,253.68
678	502	5	08	\$2,253.68
678	502	5	09	\$2,253.68
678	502	5	10	\$2,253.68
678	502	5	11	\$2,253.68
678	502	5	12	\$2,253.68
678	502	5	13	\$2,253.68
678	502	5	14	\$2,253.68
678	502	5	15	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	502	5	16	\$2,253.68
678	502	6	01	\$2,253.68
678	502	6	02	\$2,253.68
678	502	6	03	\$2,253.68
678	502	6	04	\$2,253.68
678	502	6	05	\$2,253.68
678	502	6	06	\$2,253.68
678	502	6	07	\$2,253.68
678	502	6	08	\$2,253.68
678	502	6	09	\$2,253.68
678	502	6	10	\$2,253.68
678	502	6	11	\$2,253.68
678	502	7	01	\$2,253.68
678	502	7	02	\$2,253.68
678	502	7	03	\$2,253.68
678	502	7	04	\$2,253.68
678	502	7	05	\$2,253.68
678	502	7	06	\$2,253.68
678	502	7	07	\$2,253.68
678	502	7	08	\$2,253.68
678	502	7	09	\$2,253.68
678	502	7	10	\$2,253.68
678	502	7	11	\$2,253.68
678	502	7	12	\$2,253.68
678	502	7	13	\$2,253.68
678	502	7	14	\$2,253.68
678	502	7	15	\$2,253.68
678	502	7	16	\$2,253.68
678	502	8	01	\$2,253.68
678	502	8	02	\$2,253.68
678	502	8	03	\$2,253.68
678	502	8	04	\$2,253.68
678	502	8	05	\$2,253.68
678	502	8	06	\$2,253.68
678	502	8	07	\$2,253.68
678	502	8	08	\$2,253.68
678	502	8	09	\$2,253.68
678	502	8	10	\$2,253.68
678	502	8	11	\$2,253.68
678	502	8	12	\$2,253.68
678	502	8	13	\$2,253.68
678	502	8	14	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	502	8	15	\$2,253.68
678	502	8	16	\$2,253.68
678	502	9	01	\$2,265.82
678	502	9	02	\$2,265.82
678	502	9	03	\$2,265.82
678	502	9	04	\$2,265.82
678	502	9	05	\$2,265.82
678	502	9	06	\$2,265.82
678	502	9	07	\$2,265.82
678	502	9	08	\$2,265.82
678	502	9	09	\$2,265.82
678	502	9	10	\$2,265.82
678	502	9	11	\$2,265.82
678	502	9	12	\$2,265.82
678	502	10	01	\$2,265.82
678	502	10	02	\$2,265.82
678	502	10	03	\$2,265.82
678	502	10	04	\$2,265.82
678	502	10	05	\$2,265.82
678	502	10	06	\$2,265.82
678	502	10	07	\$2,265.82
678	502	10	08	\$2,265.82
678	502	10	09	\$2,265.82
678	502	10	10	\$2,265.82
678	502	10	11	\$2,265.82
678	502	10	12	\$2,265.82
678	502	10	13	\$2,265.82
678	502	10	14	\$2,265.82
678	502	10	15	\$2,265.82
678	502	10	16	\$2,265.82
678	502	11	01	\$2,265.82
678	502	11	02	\$2,265.82
678	502	11	03	\$2,265.82
678	502	11	04	\$2,265.82
678	502	11	05	\$2,265.82
678	502	11	06	\$2,265.82
678	502	11	07	\$2,265.82
678	502	11	08	\$2,265.82
678	502	11	09	\$2,265.82
678	502	11	10	\$2,265.82
678	502	11	11	\$2,265.82
678	502	11	12	\$2,265.82

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	502	11	13	\$2,265.82	
678	502	11	14	\$2,265.82	
678	502	11	15	\$2,265.82	
678	502	11	16	\$2,265.82	
678	502	12	01	\$2,265.82	
678	502	12	02	\$2,265.82	
678	502	12	03	\$2,265.82	
678	502	12	04	\$2,265.82	
678	502	12	05	\$2,265.82	
678	502	12	06	\$2,265.82	
678	502	12	07	\$2,265.82	
678	502	12	08	\$2,265.82	
678	502	12	09	\$2,265.82	
678	502	12	10	\$2,265.82	
678	502	12	11	\$2,265.82	
678	502	13	01	\$2,265.82	
678	502	13	02	\$2,265.82	
678	502	13	03	\$2,265.82	
678	502	13	04	\$2,265.82	
678	502	13	05	\$2,265.82	
678	502	13	06	\$2,265.82	
678	502	13	07	\$2,265.82	
678	502	13	08	\$2,265.82	
678	502	13	09	\$2,265.82	
678	502	13	10	\$2,265.82	
678	502	13	11	\$2,265.82	
678	502	13	12	\$2,265.82	
678	502	13	13	\$2,265.82	
678	502	13	14	\$2,265.82	
678	502	13	15	\$2,265.82	
678	502	13	16	\$2,265.82	
678	502	14	01	\$2,253.68	
678	502	14	02	\$2,253.68	
678	502	14	03	\$2,253.68	
678	502	14	04	\$2,253.68	
678	502	14	05	\$2,265.82	
678	502	14	06	\$2,265.82	
678	502	14	07	\$2,265.82	
678	502	14	08	\$2,265.82	
678	502	14	09	\$2,265.82	
678	502	14	10	\$2,265.82	
678	502	14	11	\$2,265.82	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	502	14	12	\$2,265.82
678	502	14	13	\$2,265.82
678	502	14	14	\$2,265.82
678	502	14	15	\$2,265.82
678	502	14	16	\$2,265.82
678	503	1	1	\$2,265.82
678	503	1	2	\$2,265.82
678	503	1	3	\$2,265.82
678	503	1	4	\$2,265.82
678	503	1	5	\$2,265.82
678	503	1	6	\$2,265.82
678	503	1	7	\$2,265.82
678	503	1	8	\$2,265.82
678	503	1	9	\$2,265.82
678	503	1	10	\$2,265.82
678	503	1	11	\$2,265.82
678	503	1	12	\$2,265.82
678	503	1	13	\$2,265.82
678	503	2	1	\$2,265.82
678	503	2	2	\$2,265.82
678	503	2	3	\$2,265.82
678	503	2	4	\$2,265.82
678	503	2	5	\$2,265.82
678	503	2	6	\$2,265.82
678	503	2	7	\$2,265.82
678	503	2	8	\$2,265.82
678	503	2	9	\$2,265.82
678	503	2	10	\$2,265.82
678	503	2	11	\$2,265.82
678	503	2	12	\$2,265.82
678	503	3	1	\$2,265.82
678	503	3	2	\$2,265.82
678	503	3	3	\$2,265.82
678	503	3	4	\$2,265.82
678	503	3	5	\$2,265.82
678	503	3	6	\$2,265.82
678	503	3	7	\$2,265.82
678	503	3	8	\$2,265.82
678	503	3	9	\$2,265.82
678	503	3	10	\$2,265.82
678	503	3	11	\$2,265.82
678	503	3	12	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	503	3	13	\$2,265.82
678	503	3	14	\$2,265.82
678	503	4	1	\$2,265.82
678	503	4	2	\$2,265.82
678	503	4	3	\$2,265.82
678	503	4	4	\$2,265.82
678	503	4	5	\$2,265.82
678	503	4	6	\$2,265.82
678	503	4	7	\$2,265.82
678	503	4	8	\$2,265.82
678	503	4	9	\$2,265.82
678	503	4	10	\$2,265.82
678	503	4	11	\$2,265.82
678	503	4	12	\$2,265.82
678	503	4	13	\$2,265.82
678	503	4	14	\$2,265.82
678	503	4	15	\$2,265.82
678	610	1	01	\$2,253.68
678	610	1	02	\$2,253.68
678	610	1	03	\$2,253.68
678	610	1	04	\$2,253.68
678	610	1	05	\$2,253.68
678	610	1	06	\$2,253.68
678	610	1	07	\$2,253.68
678	610	1	08	\$2,253.68
678	610	1	09	\$2,253.68
678	610	1	10	\$2,253.68
678	610	2	01	\$2,253.68
678	610	2	02	\$2,253.68
678	610	2	03	\$2,253.68
678	610	2	04	\$2,253.68
678	610	2	05	\$2,253.68
678	610	2	06	\$2,253.68
678	610	2	07	\$2,253.68
678	610	2	08	\$2,253.68
678	610	2	09	\$2,253.68
678	610	2	10	\$2,253.68
678	610	3	01	\$2,253.68
678	610	3	02	\$2,253.68
678	610	3	03	\$2,253.68
678	610	3	04	\$2,253.68
678	610	3	05	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	610	3	06	\$2,253.68
678	610	3	07	\$2,253.68
678	610	3	08	\$2,253.68
678	610	3	09	\$2,253.68
678	610	3	10	\$2,253.68
678	610	4	01	\$2,253.68
678	610	4	02	\$2,253.68
678	610	4	03	\$2,253.68
678	610	4	04	\$2,253.68
678	610	4	05	\$2,253.68
678	610	4	06	\$2,253.68
678	610	4	07	\$2,253.68
678	610	4	08	\$2,253.68
678	610	4	09	\$2,253.68
678	610	4	10	\$2,253.68
678	610	5	01	\$2,253.68
678	610	5	02	\$2,253.68
678	610	5	03	\$2,253.68
678	610	5	04	\$2,253.68
678	610	5	05	\$2,253.68
678	610	5	06	\$2,253.68
678	610	5	07	\$2,253.68
678	610	5	08	\$2,253.68
678	610	5	09	\$2,253.68
678	610	6	01	\$2,253.68
678	610	6	02	\$2,253.68
678	610	6	03	\$2,253.68
678	610	6	04	\$2,253.68
678	610	6	05	\$2,253.68
678	610	6	06	\$2,253.68
678	610	6	07	\$2,253.68
678	610	6	08	\$2,253.68
678	610	6	09	\$2,253.68
678	610	6	10	\$2,253.68
678	610	7	01	\$2,253.68
678	610	7	02	\$2,253.68
678	610	7	03	\$2,253.68
678	610	7	04	\$2,253.68
678	610	7	05	\$2,253.68
678	610	7	06	\$2,253.68
678	610	7	07	\$2,253.68
678	610	7	08	\$2,253.68

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	610	7	09	\$2,265.82
678	610	7	10	\$2,265.82
678	610	7	11	\$2,265.82
678	610	8	01	\$2,253.68
678	610	8	02	\$2,253.68
678	610	8	03	\$2,253.68
678	610	8	04	\$2,253.68
678	610	8	05	\$2,253.68
678	610	8	06	\$2,253.68
678	610	8	07	\$2,253.68
678	610	8	08	\$2,253.68
678	610	8	09	\$2,253.68
678	610	8	10	\$2,253.68
678	610	9	01	\$2,253.68
678	610	9	02	\$2,253.68
678	610	9	03	\$2,253.68
678	610	9	04	\$2,253.68
678	610	9	05	\$2,253.68
678	610	9	06	\$2,265.82
678	610	9	07	\$2,253.68
678	610	9	08	\$2,253.68
678	610	9	09	\$2,253.68
678	610	9	10	\$2,265.82
678	610	10	01	\$2,253.68
678	610	10	02	\$2,265.82
678	610	10	03	\$2,265.82
678	610	10	04	\$2,265.82
678	610	10	05	\$2,265.82
678	610	10	06	\$2,265.82
678	610	10	07	\$2,265.82
678	610	10	08	\$2,265.82
678	610	10	09	\$2,265.82
678	610	10	10	\$2,253.68
678	610	11	01	\$2,265.82
678	610	11	02	\$2,265.82
678	610	11	03	\$2,265.82
678	610	11	04	\$2,265.82
678	610	11	05	\$2,265.82
678	610	11	06	\$2,265.82
678	610	11	07	\$2,265.82
678	610	11	08	\$2,265.82
678	610	11	09	\$2,265.82

	ASSESSOR PA	RCEL NUMBER		SPECIAL	
воок	BOOK PAGE PARCEL INT				
678	610	12	01	\$2,265.82	
678	610	12	02	\$2,265.82	
678	610	12	03	\$2,265.82	
678	610	12	04	\$2,265.82	
678	610	12	05	\$2,265.82	
678	610	12	06	\$2,265.82	
678	610	12	07	\$2,265.82	
678	610	12	08	\$2,265.82	
678	610	12	09	\$2,265.82	
678	610	12	10	\$2,265.82	
678	610	12	11	\$2,265.82	
678	610	13	01	\$2,265.82	
678	610	13	02	\$2,265.82	
678	610	13	03	\$2,265.82	
678	610	13	04	\$2,253.68	
678	610	13	05	\$2,253.68	
678	610	13	06	\$2,253.68	
678	610	13	07	\$2,265.82	
678	610	13	08	\$2,265.82	
678	610	13	09	\$2,265.82	
678	610	13	10	\$2,265.82	
678	610	13	11	\$2,265.82	
678	610	13	12	\$2,265.82	
678	610	13	13	\$2,265.82	
678	638	7	1	\$2,462.36	
678	638	7	2	\$2,462.36	
678	638	7	3	\$2,462.36	
678	638	7	4	\$2,462.36	
678	638	7	5	\$2,462.36	
678	638	7	6	\$2,462.36	
678	638	7	7	\$2,462.36	
678	638	7	8	\$2,462.36	
678	638	7	9	\$2,462.36	
678	638	7	10	\$2,462.36	
678	638	7	11	\$2,462.36	
678	638	7	12	\$2,462.36	
678	638	7	13	\$2,462.36	
678	638	7	14	\$2,462.36	
678	638	7	15	\$2,462.36	
678	638	7	16	\$2,462.36	
678	638	7	17	\$2,462.36	
678	638	7	18	\$2,462.36	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	638	7	19	\$2,462.36
678	638	7	20	\$2,462.36
678	638	7	21	\$2,462.36
678	638	7	22	\$2,462.36
678	638	7	23	\$2,462.36
678	638	7	24	\$2,462.36
678	638	7	25	\$2,462.36
678	638	7	26	\$2,462.36
678	638	7	27	\$2,462.36
678	638	7	28	\$2,462.36
678	638	7	29	\$2,462.36
678	638	7	30	\$2,462.36
678	638	7	31	\$2,462.36
678	638	7	32	\$2,462.36
678	638	7	33	\$2,462.36
678	638	7	34	\$2,462.36
678	638	7	35	\$2,462.36
678	638	7	36	\$2,462.36
678	638	7	37	\$2,462.36
678	638	7	38	\$2,462.36
678	638	7	39	\$2,462.36
678	638	7	40	\$2,462.36
678	638	7	41	\$2,462.36
678	638	7	42	\$2,462.36
678	638	7	43	\$2,462.36
678	638	7	44	\$2,462.36
678	638	7	45	\$2,462.36
678	638	7	46	\$2,462.36
678	638	8	1	\$2,462.36
678	638	8	2	\$2,462.36
678	638	8	3	\$2,462.36
678	638	8	4	\$2,462.36
678	638	8	5	\$2,462.36
678	638	8	6	\$2,462.36
678	638	8	7	\$2,462.36
678	638	8	8	\$2,462.36
678	638	8	9	\$2,462.36
678	638	8	10	\$2,462.36
678	638	8	11	\$2,462.36
678	638	8	12	\$2,462.36
678	638	8	13	\$2,462.36
678	638	8	14	\$2,462.36

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	638	8	15	\$2,462.36	
678	638	8	16	\$2,462.36	
678	638	8	17	\$2,462.36	
678	638	8	18	\$2,462.36	
678	638	8	19	\$2,462.36	
678	638	8	20	\$2,462.36	
678	638	8	21	\$2,462.36	
678	638	8	22	\$2,462.36	
678	638	8	23	\$2,462.36	
678	638	8	24	\$2,462.36	
678	638	8	25	\$2,462.36	
678	638	8	26	\$2,462.36	
678	638	8	27	\$2,462.36	
678	638	8	28	\$2,462.36	
678	638	8	29	\$2,462.36	
678	638	8	30	\$2,462.36	
678	638	8	31	\$2,462.36	
678	638	8	32	\$2,462.36	
678	638	8	33	\$2,462.36	
678	638	8	34	\$2,462.36	
678	638	8	35	\$2,462.36	
678	638	8	36	\$2,462.36	
678	638	8	37	\$2,462.36	
678	638	8	38	\$2,462.36	
678	638	8	39	\$2,462.36	
678	638	8	40	\$2,462.36	
678	638	8	41	\$2,462.36	
678	638	8	42	\$2,462.36	
678	638	8	43	\$2,462.36	
678	638	8	44	\$2,462.36	
678	638	8	45	\$2,462.36	
678	638	8	46	\$0.00	
678	638	8	47	\$0.00	
678	638	8	48	\$0.00	
678	638	8	49	\$0.00	
678	638	8	50	\$0.00	
678	638	8	51	\$0.00	
678	638	8	52	\$0.00	
678	638	8	53	\$0.00	
678	638	8	54	\$0.00	
678	638	9	1	\$2,462.36	
678	638	9	2	\$2,462.36	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
BOOK	PAGE	PARCEL	INT	TAX
678	638	9	3	\$2,462.36
678	638	9	4	\$2,462.36
678	638	9	5	\$2,462.36
678	638	9	6	\$2,462.36
678	638	9	7	\$2,462.36
678	638	9	8	\$2,462.36
678	638	9	9	\$2,462.36
678	638	9	10	\$2,462.36
678	638	9	11	\$2,462.36
678	638	9	12	\$2,462.36
678	638	9	13	\$2,462.36
678	638	9	14	\$2,462.36
678	638	9	15	\$2,462.36
678	638	9	16	\$2,462.36
678	638	9	17	\$2,462.36
678	638	9	18	\$2,462.36
678	638	9	19	\$2,462.36
678	638	9	20	\$2,462.36
678	638	9	21	\$2,462.36
678	638	9	22	\$2,462.36
678	638	9	23	\$2,462.36
678	638	9	24	\$2,462.36
678	638	9	25	\$2,462.36
678	638	9	26	\$2,462.36
678	638	9	27	\$2,462.36
678	638	9	28	\$2,462.36
678	638	9	29	\$2,462.36
678	638	9	30	\$0.00
678	638	9	31	\$0.00
678	638	9	32	\$0.00
678	638	9	33	\$0.00
678	638	9	34	\$2,462.36
678	638	9	35	\$2,462.36
678	638	9	36	\$2,462.36
678	637	12	1	\$0.00
678	637	12	2	\$0.00
678	637	12	3	\$0.00
678	637	12	4	\$0.00
678	637	12	5	\$0.00
678	637	12	6	\$0.00
678	637	12	7	\$0.00
678	637	12	8	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	637	12	9	\$0.00
678	637	12	10	\$0.00
678	637	12	11	\$0.00
678	637	12	12	\$0.00
678	637	12	13	\$0.00
678	637	12	14	\$0.00
678	637	12	15	\$0.00
678	637	12	16	\$0.00
678	637	12	17	\$0.00
678	637	12	18	\$0.00
678	637	12	19	\$0.00
678	637	12	20	\$0.00
678	637	12	21	\$0.00
678	637	12	22	\$0.00
678	637	12	23	\$0.00
678	637	12	24	\$0.00
678	637	12	25	\$0.00
678	637	12	26	\$0.00
678	637	12	27	\$0.00
678	637	12	28	\$0.00
678	637	12	29	\$0.00
678	637	12	30	\$0.00
678	637	12	31	\$0.00
678	637	12	32	\$0.00
678	637	12	33	\$0.00
678	637	12	34	\$0.00
678	637	12	35	\$0.00
678	637	12	36	\$0.00
678	637	12	37	\$0.00
678	637	12	38	\$0.00
678	637	12	39	\$0.00
678	637	12	40	\$0.00
678	637	13	1	\$0.00
678	637	13	2	\$0.00
678	637	13	3	\$0.00
678	637	13	4	\$0.00
678	637	13	5	\$0.00
678	637	13	6	\$0.00
678	637	13	7	\$0.00
678	637	13	8	\$0.00
678	637	13	9	\$0.00
678	637	13	10	\$0.00

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	637	13	11	\$0.00	
678	637	13	12	\$0.00	
678	637	13	13	\$0.00	
678	637	13	14	\$0.00	
678	637	13	15	\$0.00	
678	637	13	16	\$0.00	
678	637	13	17	\$0.00	
678	637	13	18	\$0.00	
678	637	13	19	\$0.00	
678	637	13	20	\$0.00	
678	637	13	21	\$0.00	
678	637	13	22	\$0.00	
678	637	13	23	\$0.00	
678	637	13	24	\$0.00	
678	637	13	25	\$0.00	
678	637	13	26	\$0.00	
678	637	13	27	\$0.00	
678	637	13	28	\$0.00	
678	637	13	29	\$0.00	
678	637	14	1	\$0.00	
678	637	14	2	\$0.00	
678	637	14	3	\$0.00	
678	637	14	4	\$0.00	
678	637	14	5	\$0.00	
678	637	14	6	\$0.00	
678	637	14	7	\$0.00	
678	637	14	8	\$0.00	
678	637	14	9	\$0.00	
678	637	14	10	\$0.00	
678	637	14	11	\$0.00	
678	637	14	12	\$0.00	
678	637	14	13	\$0.00	
678	637	14	14	\$0.00	
678	637	14	15	\$0.00	
678	637	14	16	\$0.00	
678	637	14	17	\$0.00	
678	637	14	18	\$0.00	
678	637	14	19	\$0.00	
678	637	14	20	\$0.00	
678	637	14	21	\$0.00	
678	637	14	22	\$0.00	
678	637	14	23	\$0.00	

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	637	14	24	\$0.00	
678	637	14	25	\$0.00	
678	637	14	26	\$0.00	
678	637	14	27	\$0.00	
678	637	14	28	\$0.00	
678	637	14	29	\$0.00	
678	637	14	30	\$0.00	
678	637	14	31	\$0.00	
678	637	14	32	\$0.00	
678	637	14	33	\$0.00	
678	637	14	34	\$0.00	
678	637	14	35	\$0.00	
678	637	14	36	\$0.00	
678	637	14	37	\$0.00	
678	637	14	38	\$0.00	
678	637	14	39	\$0.00	
678	637	14	40	\$0.00	
678	637	15	1	\$0.00	
678	637	15	2	\$0.00	
678	637	15	3	\$0.00	
678	637	15	4	\$0.00	
678	637	15	5	\$0.00	
678	637	15	6	\$0.00	
678	637	15	7	\$0.00	
678	637	15	8	\$0.00	
678	637	15	9	\$0.00	
678	637	15	10	\$0.00	
678	637	15	11	\$0.00	
678	637	15	12	\$0.00	
678	637	15	13	\$0.00	
678	637	15	14	\$0.00	
678	637	15	15	\$0.00	
678	637	15	16	\$0.00	
678	637	15	17	\$0.00	
678	637	15	18	\$0.00	
678	637	15	19	\$0.00	
678	637	15	20	\$0.00	
678	637	15	21	\$0.00	
678	637	15	22	\$0.00	
678	637	15	23	\$0.00	
678	637	15	24	\$0.00	
678	637	15	25	\$0.00	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	637	15	26	\$0.00
678	637	15	27	\$0.00
678	637	15	28	\$0.00
678	637	15	29	\$0.00
678	637	16	1	\$0.00
678	637	16	2	\$0.00
678	637	16	3	\$0.00
678	637	16	4	\$0.00
678	637	16	5	\$0.00
678	637	16	6	\$0.00
678	637	16	7	\$0.00
678	637	16	8	\$0.00
678	637	16	9	\$0.00
678	637	16	10	\$0.00
678	637	16	11	\$0.00
678	637	16	12	\$0.00
678	637	16	13	\$0.00
678	637	16	14	\$0.00
678	637	16	15	\$0.00
678	637	16	16	\$0.00
678	637	16	17	\$0.00
678	637	16	18	\$0.00
678	637	16	19	\$0.00
678	637	16	20	\$0.00
678	637	16	21	\$0.00
678	637	16	22	\$0.00
678	637	16	23	\$0.00
678	637	16	24	\$0.00
678	637	16	25	\$0.00
678	637	16	26	\$0.00
678	637	16	27	\$0.00
678	637	16	28	\$0.00
678	637	16	29	\$0.00
678	637	16	30	\$0.00
678	637	16	31	\$0.00
678	637	16	32	\$0.00
678	637	16	33	\$0.00
678	637	16	34	\$0.00
678	637	16	35	\$0.00
678	637	16	36	\$0.00
678	637	16	37	\$0.00
678	637	16	38	\$0.00

	ASSESSOR PARCEL NUMBER					
BOOK	PAGE	PARCEL	INT	TAX		
678	637	16	39	\$0.00		
678	637	16	40	\$0.00		
678	637	17	1	\$0.00		
678	637	17	2	\$0.00		
678	637	17	3	\$0.00		
678	637	17	4	\$0.00		
678	637	17	5	\$0.00		
678	637	17	6	\$0.00		
678	637	17	7	\$0.00		
678	637	17	8	\$0.00		
678	637	17	9	\$0.00		
678	637	17	10	\$0.00		
678	637	17	11	\$0.00		
678	637	17	12	\$0.00		
678	637	17	13	\$0.00		
678	637	17	14	\$0.00		
678	637	17	15	\$0.00		
678	637	17	16	\$0.00		
678	637	17	17	\$0.00		
678	637	17	18	\$0.00		
678	637	17	19	\$0.00		
678	637	17	20	\$0.00		
678	637	17	21	\$0.00		
678	637	17	22	\$0.00		
678	637	17	23	\$0.00		
678	637	17	24	\$0.00		
678	637	17	25	\$0.00		
678	637	17	26	\$0.00		
678	637	17	27	\$0.00		
678	637	17	28	\$0.00		
678	637	17	29	\$0.00		
678	637	17	30	\$0.00		
678	637	17	31	\$0.00		
678	637	17	32	\$0.00		
678	637	17	33	\$0.00		
678	637	17	34	\$0.00		
678	637	17	35	\$0.00		
678	637	17	36	\$0.00		
678	637	17	37	\$0.00		
678	637	17	38	\$0.00		
678	637	17	39	\$0.00		
678	637	17	40	\$0.00		

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	637	10	01	\$1,089.12	
678	637	10	02	\$1,089.12	
678	637	10	03	\$1,089.12	
678	637	10	04	\$1,089.12	
678	637	10	05	\$1,089.12	
678	637	10	06	\$1,089.12	
678	637	10	07	\$1,089.12	
678	637	10	08	\$1,089.12	
678	637	10	09	\$1,089.12	
678	637	10	10	\$1,089.12	
678	637	10	11	\$1,089.12	
678	637	10	12	\$1,089.12	
678	637	10	13	\$1,089.12	
678	637	10	14	\$1,089.12	
678	637	10	15	\$1,089.12	
678	637	10	16	\$1,089.12	
678	637	10	17	\$1,089.12	
678	637	10	18	\$1,089.12	
678	637	10	19	\$1,089.12	
678	637	10	20	\$1,089.12	
678	637	10	21	\$1,089.12	
678	637	10	22	\$1,089.12	
678	637	10	23	\$1,089.12	
678	637	10	24	\$1,089.12	
678	637	10	25	\$1,089.12	
678	637	10	26	\$1,089.12	
678	637	10	27	\$1,089.12	
678	637	10	28	\$1,089.12	
678	637	10	29	\$1,089.12	
678	637	10	30	\$1,089.12	
678	637	10	31	\$1,089.12	
678	637	10	32	\$1,089.12	
678	637	10	33	\$1,089.12	
678	637	10	34	\$1,089.12	
678	637	10	35	\$1,089.12	
678	637	10	36	\$1,089.12	
678	637	10	37	\$1,089.12	
678	637	10	38	\$1,089.12	
678	637	10	39	\$1,089.12	
678	637	10	40	\$1,089.12	
678	637	10	41	\$1,089.12	
678	637	10	42	\$1,089.12	

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	637	10	43	\$1,089.12	
678	637	10	44	\$1,089.12	
678	637	10	45	\$1,089.12	
678	637	10	46	\$1,089.12	
678	637	10	47	\$1,089.12	
678	637	10	48	\$1,089.12	
678	637	10	49	\$1,089.12	
678	637	10	50	\$1,089.12	
678	637	10	51	\$1,089.12	
678	637	10	52	\$1,089.12	
678	637	10	53	\$1,089.12	
678	637	10	54	\$1,089.12	
678	637	10	55	\$1,089.12	
678	637	10	56	\$1,089.12	
678	637	10	57	\$1,089.12	
678	637	10	58	\$1,089.12	
678	637	10	59	\$1,089.12	
678	637	10	60	\$1,089.12	
678	637	10	61	\$1,089.12	
678	637	10	62	\$1,089.12	
678	637	10	63	\$1,089.12	
678	637	10	64	\$1,089.12	
678	637	10	65	\$1,089.12	
678	637	10	66	\$1,089.12	
678	637	10	67	\$1,089.12	
678	637	10	68	\$1,089.12	
678	637	10	69	\$1,089.12	
678	637	10	70	\$1,089.12	
678	637	10	71	\$1,089.12	
678	637	10	72	\$1,089.12	
678	637	11	01	\$1,089.12	
678	637	11	02	\$1,089.12	
678	637	11	03	\$1,089.12	
678	637	11	04	\$1,089.12	
678	637	11	05	\$1,089.12	
678	637	11	06	\$1,089.12	
678	637	11	07	\$1,089.12	
678	637	11	08	\$1,089.12	
678	637	11	09	\$1,089.12	
678	637	11	10	\$1,089.12	
678	637	11	11	\$1,089.12	
678	637	11	12	\$1,089.12	

	SPECIAL				
воок	BOOK PAGE PARCEL INT				
678	637	11	13	\$1,089.12	
678	637	11	14	\$1,089.12	
678	637	11	15	\$1,089.12	
678	637	11	16	\$1,089.12	
678	637	11	17	\$1,089.12	
678	637	11	18	\$1,089.12	
678	637	11	19	\$1,089.12	
678	637	11	20	\$1,089.12	
678	637	11	21	\$1,089.12	
678	637	11	22	\$1,089.12	
678	637	11	23	\$1,089.12	
678	637	11	24	\$1,089.12	
678	637	11	25	\$1,089.12	
678	637	11	26	\$1,089.12	
678	637	11	27	\$1,089.12	
678	637	11	28	\$1,089.12	
678	637	11	29	\$1,089.12	
678	637	11	30	\$1,089.12	
678	637	11	31	\$1,089.12	
678	637	11	32	\$1,089.12	
678	637	11	33	\$1,089.12	
678	637	11	34	\$1,089.12	
678	637	11	35	\$1,089.12	
678	637	11	36	\$1,089.12	
678	637	11	37	\$1,089.12	
678	637	11	38	\$1,089.12	
678	637	11	39	\$1,089.12	
678	637	11	40	\$1,089.12	
678	637	11	41	\$1,089.12	
678	637	11	42	\$1,089.12	
678	637	11	43	\$1,089.12	
678	637	11	44	\$1,089.12	
678	637	11	45	\$1,089.12	
678	637	11	46	\$1,089.12	
678	637	11	47	\$1,089.12	
678	637	11	48	\$1,089.12	
678	637	11	49	\$1,089.12	
678	637	11	50	\$1,089.12	
678	637	11	51	\$1,089.12	
678	637	11	52	\$1,089.12	
678	637	11	53	\$1,089.12	
678	637	11	54	\$1,089.12	

	ASSESSOR PA	RCEL NUMBER		SPECIAL	
воок	BOOK PAGE PARCEL INT				
678	637	11	55	\$1,089.12	
678	637	11	56	\$1,089.12	
678	637	11	57	\$1,089.12	
678	637	11	58	\$1,089.12	
678	637	11	59	\$1,089.12	
678	637	11	60	\$1,089.12	
678	637	11	61	\$1,089.12	
678	637	11	62	\$1,089.12	
678	637	11	63	\$1,089.12	
678	637	11	64	\$1,089.12	
678	637	11	65	\$1,089.12	
678	637	11	66	\$1,089.12	
678	637	11	67	\$1,089.12	
678	637	11	68	\$1,089.12	
678	638	02	00	\$0.00	
678	660	1	00	\$0.00	
678	660	2	00	\$0.00	
678	660	3	00	\$0.00	
678	660	4	00	\$0.00	
678	660	5	00	\$0.00	
678	660	6	00	\$0.00	
678	660	7	00	\$0.00	
678	660	8	00	\$0.00	
678	660	9	00	\$0.00	
678	660	10	00	\$0.00	
678	660	11	00	\$0.00	
678	660	12	00	\$0.00	
678	660	13	00	\$0.00	
678	660	14	00	\$0.00	
678	660	15	00	\$0.00	
678	660	16	00	\$0.00	
678	660	17	00	\$0.00	
678	660	18	00	\$0.00	
678	660	19	00	\$0.00	
678	660	20	00	\$0.00	
678	660	21	00	\$0.00	
678	660	22	00	\$0.00	
678	660	23	00	\$0.00	
678	660	24	00	\$0.00	
678	660	25	00	\$0.00	
678	660	26	00	\$0.00	
678	660	27	00	\$0.00	

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	660	28	00	\$0.00	
678	660	29	00	\$0.00	
678	660	30	00	\$0.00	
678	660	31	00	\$0.00	
678	660	32	00	\$0.00	
678	660	33	00	\$0.00	
678	660	34	00	\$0.00	
678	660	35	00	\$0.00	
678	660	36	00	\$0.00	
678	660	37	00	\$0.00	
678	660	38	00	\$0.00	
678	660	39	00	\$0.00	
678	660	40	00	\$0.00	
678	660	41	00	\$0.00	
678	660	42	00	\$0.00	
678	660	43	00	\$0.00	
678	660	44	00	\$0.00	
678	660	45	00	\$0.00	
678	660	46	00	\$0.00	
678	660	47	00	\$0.00	
678	660	48	00	\$0.00	
678	660	49	00	\$0.00	
678	660	50	00	\$0.00	
678	660	51	00	\$0.00	
678	660	52	00	\$0.00	
678	660	53	00	\$0.00	
678	660	54	00	\$0.00	
678	660	55	00	\$0.00	
678	660	56	00	\$0.00	
678	660	57	00	\$0.00	
678	660	58	00	\$0.00	
678	660	59	00	\$0.00	
678	660	60	00	\$0.00	
678	660	61	00	\$0.00	
678	660	62	00	\$0.00	
678	660	63	00	\$0.00	
678	660	64	00	\$0.00	
678	660	65	00	\$0.00	
678	660	66	00	\$0.00	
678	660	67	00	\$0.00	
678	660	68	00	\$0.00	
678	660	69	00	\$0.00	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	BOOK PAGE PARCEL INT			
678	660	70	00	\$0.00
678	660	71	00	\$0.00
678	660	72	00	\$0.00
678	660	73	00	\$0.00
678	660	74	00	\$0.00
678	660	75	00	\$0.00
678	660	76	00	\$0.00
678	661	1	00	\$0.00
678	661	2	00	\$0.00
678	661	3	00	\$0.00
678	661	4	00	\$0.00
678	661	5	00	\$0.00
678	661	6	00	\$0.00
678	661	7	00	\$0.00
678	661	8	00	\$0.00
678	661	9	00	\$0.00
678	661	10	00	\$0.00
678	661	11	00	\$0.00
678	661	12	00	\$0.00
678	661	13	00	\$0.00
678	661	14	00	\$0.00
678	661	15	00	\$0.00
678	661	16	00	\$0.00
678	661	17	00	\$0.00
678	661	18	00	\$0.00
678	661	19	00	\$0.00
678	661	20	00	\$0.00
678	661	21	00	\$0.00
678	661	22	00	\$0.00
678	661	23	00	\$0.00
678	661	24	00	\$0.00
678	661	25	00	\$0.00
678	661	26	00	\$0.00
678	661	27	00	\$0.00
678	661	28	00	\$0.00
678	661	29	00	\$0.00
678	661	30	00	\$0.00
678	661	31	00	\$0.00
678	661	32	00	\$0.00
678	661	33	00	\$0.00
678	661	34	00	\$0.00
678	661	35	00	\$0.00

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	BOOK PAGE PARCEL INT			
678	661	36	00	\$0.00
678	661	37	00	\$0.00
678	661	38	00	\$0.00
678	661	39	00	\$0.00
678	661	40	00	\$0.00
678	661	41	00	\$0.00
678	661	42	00	\$0.00
678	661	43	00	\$0.00
678	661	44	00	\$0.00
678	661	45	00	\$0.00
678	661	46	00	\$0.00
678	661	47	00	\$0.00
678	661	48	00	\$0.00
678	661	49	00	\$0.00
678	662	1	00	\$0.00
678	662	2	00	\$0.00
678	662	3	00	\$0.00
678	662	4	00	\$0.00
678	662	5	00	\$0.00
678	662	6	00	\$0.00
678	662	7	00	\$0.00
678	662	8	00	\$0.00
678	662	9	00	\$0.00
678	662	10	00	\$0.00
678	662	11	00	\$0.00
678	662	12	00	\$0.00
678	662	13	00	\$0.00
678	662	14	00	\$0.00
678	662	15	00	\$0.00
678	662	16	00	\$0.00
678	662	17	00	\$0.00
678	662	18	00	\$0.00
678	662	19	00	\$0.00
678	662	20	00	\$0.00
678	662	21	00	\$0.00
678	662	22	00	\$0.00
678	662	23	00	\$0.00
678	662	24	00	\$0.00
678	662	25	00	\$0.00
678	662	26	00	\$0.00
678	662	27	00	\$0.00
678	662	28	00	\$0.00

	ASSESSOR PARCEL NUMBER					
воок	PAGE	PARCEL	INT	TAX		
678	662	29	00	\$0.00		
678	662	30	00	\$0.00		
678	662	31	00	\$0.00		
678	662	32	00	\$0.00		
678	662	33	00	\$0.00		
678	662	34	00	\$0.00		
678	662	35	00	\$0.00		
678	662	36	00	\$0.00		
678	662	37	00	\$0.00		
678	662	38	00	\$0.00		
678	662	39	00	\$0.00		
678	662	40	00	\$0.00		
678	662	41	00	\$0.00		
678	662	42	00	\$0.00		
678	662	43	00	\$0.00		
678	662	44	00	\$0.00		
678	662	45	00	\$0.00		
678	662	46	00	\$0.00		
678	662	47	00	\$0.00		
678	662	48	00	\$0.00		
678	662	49	00	\$0.00		
678	662	50	00	\$0.00		
678	662	51	00	\$0.00		
678	662	52	00	\$0.00		
678	662	53	00	\$0.00		
678	662	54	00	\$0.00		
678	662	55	00	\$0.00		
678	662	56	00	\$0.00		
678	662	57	00	\$0.00		
678	662	58	00	\$0.00		
678	662	59	00	\$0.00		
678	662	60	00	\$0.00		
678	662	61	00	\$0.00		
678	662	62	00	\$0.00		
678	662	63	00	\$0.00		
678	662	64	00	\$0.00		
678	662	65	00	\$0.00		
678	662	66	00	\$0.00		
678	662	67	00	\$0.00		
678	662	68	00	\$0.00		
678	662	69	00	\$0.00		
678	662	70	00	\$0.00		

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	662	71	00	\$0.00	
678	662	72	00	\$0.00	
678	662	73	00	\$0.00	
678	662	74	00	\$0.00	
678	662	75	00	\$0.00	
678	662	76	00	\$0.00	
678	663	1	00	\$0.00	
678	663	2	00	\$0.00	
678	663	3	00	\$0.00	
678	663	4	00	\$0.00	
678	663	5	00	\$0.00	
678	663	6	00	\$0.00	
678	663	7	00	\$0.00	
678	663	8	00	\$0.00	
678	663	9	00	\$0.00	
678	663	10	00	\$0.00	
678	663	11	00	\$0.00	
678	663	12	00	\$0.00	
678	663	13	00	\$0.00	
678	663	14	00	\$0.00	
678	663	15	00	\$0.00	
678	663	16	00	\$0.00	
678	663	17	00	\$0.00	
678	663	18	00	\$0.00	
678	663	19	00	\$0.00	
678	663	20	00	\$0.00	
678	663	21	00	\$0.00	
678	663	22	00	\$0.00	
678	663	23	00	\$0.00	
678	663	24	00	\$0.00	
678	663	25	00	\$0.00	
678	663	26	00	\$0.00	
678	663	27	00	\$0.00	
678	663	28	00	\$0.00	
678	663	29	00	\$0.00	
678	663	30	00	\$0.00	
678	663	31	00	\$0.00	
678	663	32	00	\$0.00	
678	663	33	00	\$0.00	
678	663	34	00	\$0.00	
678	663	35	00	\$0.00	
678	663	36	00	\$0.00	

	ASSESSOR PA	RCEL NUMBER		SPECIAL
воок	OOK PAGE PARCEL INT			
678	663	37	00	\$0.00
678	663	38	00	\$0.00
678	663	39	00	\$0.00
678	663	40	00	\$0.00
678	663	41	00	\$0.00
678	663	42	00	\$0.00
678	663	43	00	\$0.00
678	663	44	00	\$0.00
678	664	1	00	\$0.00
678	664	2	00	\$0.00
678	664	3	00	\$0.00
678	664	4	00	\$0.00
678	664	5	00	\$0.00
678	664	6	00	\$0.00
678	664	7	00	\$0.00
678	664	8	00	\$0.00
678	664	9	00	\$0.00
678	664	10	00	\$0.00
678	664	11	00	\$0.00
678	664	12	00	\$0.00
678	664	13	00	\$0.00
678	664	14	00	\$0.00
678	664	15	00	\$0.00
678	664	16	00	\$0.00
678	664	17	00	\$0.00
678	664	18	00	\$0.00
678	664	19	00	\$0.00
678	664	20	00	\$0.00
678	664	21	00	\$0.00
678	664	22	00	\$0.00
678	664	23	00	\$0.00
678	664	24	00	\$0.00
678	664	25	00	\$0.00
678	664	26	00	\$0.00
678	664	27	00	\$0.00
678	664	28	00	\$0.00
678	664	29	00	\$0.00
678	664	30	00	\$0.00
678	664	31	00	\$0.00
678	664	32	00	\$0.00
678	664	33	00	\$0.00
678	664	34	00	\$0.00

	SPECIAL			
воок	PAGE	PARCEL	INT	TAX
678	664	35	00	\$0.00
678	664	36	00	\$0.00
678	664	37	00	\$0.00
678	664	38	00	\$0.00
678	664	39	00	\$0.00
678	664	40	00	\$0.00
678	664	41	00	\$0.00
678	664	42	00	\$0.00
678	664	43	00	\$0.00
678	664	44	00	\$0.00
678	664	45	00	\$0.00
678	664	46	00	\$0.00
678	664	47	00	\$0.00
678	664	48	00	\$0.00
678	664	49	00	\$0.00
678	664	50	00	\$0.00
678	664	51	00	\$0.00
678	664	52	00	\$0.00
678	665	1	00	\$0.00
678	665	2	00	\$0.00
678	665	3	00	\$0.00
678	665	4	00	\$0.00
678	665	5	00	\$0.00
678	665	6	00	\$0.00
678	665	7	00	\$0.00
678	665	8	00	\$0.00
678	665	9	00	\$0.00
678	665	10	00	\$0.00
678	665	11	00	\$0.00
678	665	12	00	\$0.00
678	665	13	00	\$0.00
678	665	14	00	\$0.00
678	665	15	00	\$0.00
678	665	16	00	\$0.00
678	665	17	00	\$0.00
678	665	18	00	\$0.00
678	665	19	00	\$0.00
678	665	20	00	\$0.00
678	665	21	00	\$0.00
678	665	22	00	\$0.00
678	665	23	00	\$0.00
678	665	24	00	\$0.00

ASSESSOR PARCEL NUMBER				SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	665	25	00	\$0.00
678	665	26	00	\$0.00
678	665	27	00	\$0.00
678	665	28	00	\$0.00
678	665	29	00	\$0.00
678	665	30	00	\$0.00
678	665	31	00	\$0.00
678	665	32	00	\$0.00
678	665	33	00	\$0.00
678	665	34	00	\$0.00
678	665	35	00	\$0.00
678	665	36	00	\$0.00
678	665	37	00	\$0.00
678	665	38	00	\$0.00
678	665	39	00	\$0.00
678	665	40	00	\$0.00
678	665	41	00	\$0.00
678	665	42	00	\$0.00
678	665	43	00	\$0.00
678	665	44	00	\$0.00
678	665	45	00	\$0.00
678	665	46	00	\$0.00
678	665	47	00	\$0.00
678	665	48	00	\$0.00
678	665	49	00	\$0.00
678	665	50	00	\$0.00
678	665	51	00	\$0.00
678	665	52	00	\$0.00
678	665	53	00	\$0.00
678	665	54	00	\$0.00
678	665	55	00	\$0.00
678	665	56	00	\$0.00
678	665	57	00	\$0.00
678	665	58	00	\$0.00
678	665	59	00	\$0.00
678	665	60	00	\$0.00
678	666	1	00	\$0.00
678	666	2	00	\$0.00
678	666	3	00	\$0.00
678	666	4	00	\$0.00
678	666	5	00	\$0.00
678	666	6	00	\$0.00

	SPECIAL			
воок	PAGE	PARCEL	INT	TAX
678	666	7	00	\$0.00
678	666	8	00	\$0.00
678	666	9	00	\$0.00
678	666	10	00	\$0.00
678	666	11	00	\$0.00
678	666	12	00	\$0.00
678	666	13	00	\$0.00
678	666	14	00	\$0.00
678	666	15	00	\$0.00
678	666	16	00	\$0.00
678	666	17	00	\$0.00
678	666	18	00	\$0.00
678	666	19	00	\$0.00
678	666	20	00	\$0.00
678	666	21	00	\$0.00
678	666	22	00	\$0.00
678	666	23	00	\$0.00
678	666	24	00	\$0.00
678	666	25	00	\$0.00
678	666	26	00	\$0.00
678	666	27	00	\$0.00
678	666	28	00	\$0.00
678	666	29	00	\$0.00
678	666	30	00	\$0.00
678	666	31	00	\$0.00
678	666	32	00	\$0.00
678	666	33	00	\$0.00
678	666	34	00	\$0.00
678	666	35	00	\$0.00
678	666	36	00	\$0.00
678	666	37	00	\$0.00
678	666	38	00	\$0.00
678	666	39	00	\$0.00
678	666	40	00	\$0.00
678	666	41	00	\$0.00
678	667	1	00	\$0.00
678	667	2	00	\$0.00
678	667	3	00	\$0.00
678	667	4	00	\$0.00
678	667	5	00	\$0.00
678	667	6	00	\$0.00
678	667	7	00	\$0.00

	SPECIAL			
воок	PAGE	PARCEL	INT	TAX
678	667	8	00	\$0.00
678	667	9	00	\$0.00
678	667	10	00	\$0.00
678	667	11	00	\$0.00
678	667	12	00	\$0.00
678	667	13	00	\$0.00
678	667	14	00	\$0.00
678	667	15	00	\$0.00
678	667	16	00	\$0.00
678	667	17	00	\$0.00
678	667	18	00	\$0.00
678	667	19	00	\$0.00
678	667	20	00	\$0.00
678	667	21	00	\$0.00
678	667	22	00	\$0.00
678	667	23	00	\$0.00
678	667	24	00	\$0.00
678	667	25	00	\$0.00
678	667	26	00	\$0.00
678	667	27	00	\$0.00
678	667	28	00	\$0.00
678	667	29	00	\$0.00
678	667	30	00	\$0.00
678	667	31	00	\$0.00
678	667	32	00	\$0.00
678	667	33	00	\$0.00
678	667	34	00	\$0.00
678	667	35	00	\$0.00
678	667	36	00	\$0.00
678	667	37	00	\$0.00
678	667	38	00	\$0.00
678	667	39	00	\$0.00
678	667	40	00	\$0.00
678	667	41	00	\$0.00
678	667	42	00	\$0.00
678	667	43	00	\$0.00
678	667	44	00	\$0.00
678	667	45	00	\$0.00
678	667	46	00	\$0.00
678	667	47	00	\$0.00
678	667	48	00	\$0.00
678	667	49	00	\$0.00

	ASSESSOR PARCEL NUMBER			
воок	PAGE	PARCEL	INT	TAX
678	667	50	00	\$0.00
678	667	51	00	\$0.00
678	667	52	00	\$0.00
678	667	53	00	\$0.00
678	667	54	00	\$0.00
678	667	55	00	\$0.00
678	668	1	00	\$0.00
678	668	2	00	\$0.00
678	668	3	00	\$0.00
678	668	4	00	\$0.00
678	668	5	00	\$0.00
678	668	6	00	\$0.00
678	668	7	00	\$0.00
678	668	8	00	\$0.00
678	668	9	00	\$0.00
678	668	10	00	\$0.00
678	668	11	00	\$0.00
678	668	12	00	\$0.00
678	668	13	00	\$0.00
678	668	14	00	\$0.00
678	668	15	00	\$0.00
678	668	16	00	\$0.00
678	668	17	00	\$0.00
678	668	18	00	\$0.00
678	668	19	00	\$0.00
678	668	20	00	\$0.00
678	668	21	00	\$0.00
678	668	22	00	\$0.00
678	668	23	00	\$0.00
678	668	24	00	\$0.00
678	668	25	00	\$0.00
678	668	26	00	\$0.00
678	668	27	00	\$0.00
678	668	28	00	\$0.00
678	668	29	00	\$0.00
678	668	30	00	\$0.00
678	668	31	00	\$0.00
678	668	32	00	\$0.00
678	668	33	00	\$0.00
678	668	34	00	\$0.00
678	668	35	00	\$0.00
678	668	36	00	\$0.00

ASSESSOR PARCEL NUMBER				SPECIAL
воок	PAGE	PARCEL	INT	TAX
678	668	37	00	\$0.00
678	668	38	00	\$0.00
678	668	39	00	\$0.00
678	670	1	00	\$0.00
678	670	2	00	\$0.00
678	670	3	00	\$0.00
678	670	4	00	\$0.00
678	670	5	00	\$0.00
678	670	6	00	\$0.00
678	670	7	00	\$0.00
678	670	8	00	\$0.00
678	670	9	00	\$0.00
678	670	10	00	\$0.00
678	670	11	00	\$0.00
678	670	12	00	\$0.00
678	670	13	00	\$0.00
678	670	14	00	\$0.00
678	670	15	00	\$0.00
678	670	16	00	\$0.00
678	670	17	00	\$0.00
678	670	18	00	\$0.00
678	670	19	00	\$0.00
678	670	20	00	\$0.00
678	670	21	00	\$0.00
678	671	1	00	\$0.00
678	671	2	00	\$0.00
678	671	3	00	\$0.00
678	650	1	00	\$0.00
678	650	2	00	\$0.00
678	650	3	00	\$0.00
678	650	4	00	\$0.00
678	650	5	00	\$0.00
312	141	7	00	\$0.00
312	150	6	00	\$0.00
312	150	7	00	\$0.00
312	150	4	00	\$0.00
313	301	33	00	\$0.00
678	030	6	00	\$0.00
678	030	7	00	\$0.00
678	050	45	00	\$0.00
678	050	48	00	\$0.00
678	050	47	00	\$0.00

	ASSESSOR PARCEL NUMBER				
воок	PAGE	PARCEL	INT	TAX	
678	230	12	00	\$0.00	
678	230	28	00	\$0.00	
678	241	6	00	\$0.00	
678	242	13	00	\$0.00	
678	242	15	00	\$0.00	
678	31	1	00	\$0.00	
678	500	3	00	\$0.00	
678	501	4	00	\$0.00	
678	501	5	00	\$0.00	
678	501	6	00	\$0.00	
678	501	14	00	\$0.00	
678	501	15	00	\$0.00	
678	501	16	00	\$0.00	
678	501	8	00	\$0.00	
678	501	9	00	\$0.00	
678	600	1	00	\$0.00	
678	600	2	00	\$0.00	
678	600	3	00	\$0.00	
678	600	4	00	\$0.00	
678	600	5	00	\$0.00	
678	600	6	00	\$0.00	

MAJOR CONCLUSIONS

NUMBER OF PARCELS WITHIN CFD NO. 6	3,759
NUMBER OF PARCELS TAXED	2,455
TOTAL TAX LEVY FY 2006-07	\$5,256,325.64